Solar Energy xxx (xxxx) xxx



Contents lists available at ScienceDirect

Solar Energy



journal homepage: www.elsevier.com/locate/solener

Applications of solar energy based drying technologies in various industries – A review

Abhay Lingayat^a, Ramakrishna Balijepalli^b, V.P. Chandramohan^{c,*}

^a Department of Energy Science and Engineering, Indian Institute of Technology, Bombay, Maharashtra 400076, India

^b Department of Mechanical Engineering, RGM College of Engineering and Technology, Nandyal, Andhrapradesh 518501, India

^c Department of Mechanical Engineering, National Institute of Technology Warangal, Warangal, Telangana State 506 004, India

ARTICLE INFO

Keywords: Industrial drying Food drying Solar drying Solar energy Drying technologies

ABSTRACT

Drying of materials is an important and essential mechanism during the production process in industries. Drying fruits and vegetables is an ancient method used for food preservation. Apart from the agricultural sector, there are many industries such as automobile, rubber, paper and pulp, sugarcane, wastewater treatment, lignite/coal, etc. require heat energy for drying during production processes. Conventional energy sources such as coal, natural gas, and electricity are used for the heat energy required for drying. Due to the increase in cost and pollution involved in conventional sources, solar energy-based drying systems can be encouraged. This review work provides a detailed analysis of solar-based dryers used in various industries namely agricultural, marine, tea, sugarcane, automobile, rubber, pulp, and paper industries. In addition, the utilization of solar energy for sewage drying, industrial waste drying, and lignite coal drying for power generation are reviewed. Different types of dryers presently available in the market are also discussed in the review work. Various parameters used for the performance of the dryers such as drying rate, amount of energy required for drying, collector efficiency, drying efficiency, specific energy consumption have been discussed. Economic, environmental, and social aspects of solar dryers are also presented, and recommendations are given in the paper.

1. Introduction

In the present scenario, energy security is one of the important areas where the world is continuously looking for various methods and technologies. The main motive of energy researchers is, reducing the consumption of energy and finding alternate sources. Encouraging renewable energies as alternative options reduce hydrocarbon and other toxic emissions. Drying is one of the energy intensive processes which is used in many industries such as agriculture, sugarcane, tea, marine, automobile, paper, rubber, pulp and paper, sewage and industrial waste, lignite and coal, etc. Drying played an important role in the evaporation of moisture from industrial waste and sewage which can be used as fertilizer or for landfilling purposes (Mathioudakis et al., 2009).

The agricultural sector is a major sector where drying has been extensively used in preserving cereals, fruits and vegetables. Moisture is the main substance in the wet materials that causes the microbial and bacterial reaction which leads the material spoilage. The reduction of the moisture content (MC) up to a safe level, reduces the growth and reproduction of micro-organisms and it can be achievable only by drying. Thermal energy is a source mostly used for the drying process. Most of the time, conventional energy sources such as fossil fuel and electric energy were used for drying. Varieties of dryers available in the market with different heat supply modes and energy sources as listed in Table 1.

Consumption of fossil fuels greatly influences the environment as they emit toxic gases. The heat energy consumed during the drying process is about 12% to 40% of total industrial energy consumption in the developed countries, which employs 20–70% of the total cost of production depends on the type of industries. (Bennamoun and Belhamri, 2003; Pirasteh et al., 2014a). A natural source - solar energy could play a major role in drying processes and it minimizes the consumption of non-renewable sources by 27% to 80% (Prakash et al., 2018). Therefore, solar energy needs to be encouraged for drying applications. It is a clean, sustainable, economical, and environmentally friendly energy source.

Open sun drying (OSD) is a traditional method to dry crops, fruits, vegetables, and other products since ancient times. But it has some disadvantages such as large surface area, low quality on the dried product due to contamination of dirt and dust, unavailability of required

* Corresponding author.

E-mail address: vpcm80@nitw.ac.in (V.P. Chandramohan).

https://doi.org/10.1016/j.solener.2021.05.058

Received 16 March 2021; Received in revised form 16 May 2021; Accepted 19 May 2021 0038-092X/© 2021 International Solar Energy Society. Published by Elsevier Ltd. All rights reserved.

Nomenclature		SEC	Specific energy consumption (kJ/kg)
		TES	Thermal energy storage
Abbreviations			Temperature (K)
Α	Area of collector (m ²)	T_d	Drying temperature (K)
$C_{pa\setminus}$	Specific heat (J/ kg K)	t _s	Save in time
db	Dry basis	wb	Wet basis
DSD	Direct solar dryer		
Ex	Exergy	Greeks	
HSD	Hybrid solar dryer	η_o	Collector efficiency
H_{fuel}	Heat value of fuel (kW/ kg)	η_d	Drying efficiency
ĥ	Specific humidity (kg/ kg of dry air)	Subscrip	ts
h _{fg}	Enthalpy of evaporation (kJ/kg)	a	air
h _{as}	Specific humidity as saturated condition (kg/ kg of dry air)	atm	Atmospheric
ISD	Indirect solar dryer	c	collector
Ι	Incident solar radiation (W/m ²)	d	Drv. drver
IP	Improvement potential	dci	Drving chamber inlet
М, т	Mass (kg)	dco	Drving chamber outlet
MC	Moisture content (kg per kg)	do	Drver outlet
ṁ	Mass flow rate (kg/s)	f	final
m_m	Mass of material (kg)	i	Initial, inlet
OSD	Open sun drying	1	loss
P_t	Total energy input (kW)	in	inlet
PCM	Phase change material	m	Material
P_f	Fan power	w	water
Q_d	Energy for drying (kJ)		
SAC	Solar air collector		

Table 1

Various methods involve in the drying operation.

Category	Method and range
Mode of heat supply	Conduction, convection, radiation, infrared, dielectric
Energy source	Electricity, coal, oil, natural gas, biomass, solar energy
Moisture content in the material	1 to 96 kg per kg of wet basis
Drying temperature	30 °C – 200 °C
Pressure	Vacuum to high-pressure level
Capacity	kilogram to tones of production
Shape, size, and structure	Powder, granules, film, solid material, crystalline, fabric, cardboard, fiber, etc.

drying temperature, larger drying time for high MC product, contamination by birds and animals, etc. Varieties of other solar-based drying systems are available which can overcome the negatives of the OSD method (and therefore, the OSD method is not discussed much in this review). These methods provide the drying process more efficiently with high-quality dried products.

In the area of solar drying, many experimental, theoretical, and numerical studies are available which shows the importance of solarbased drying processes in agricultural as well as non-agricultural sectors. Also, many reviews were reported by researchers on different methods of drying in the agricultural sector (Fudholi et al., 2015; Panwar et al., 2012). There were few review works on software tools used for drying problems (Malekjani and Jafari, 2018; Singh et al., 2015).

However, despite having the huge potential of solar thermal energy, there is a lack of review work focusing on the integration of solar thermal energy with industrial drying process applications. No review work was solely contributed to solar drying processes used in different industries. Solar dryers are commonly used in the agricultural sector which is mentioned in lots of studies but the available review works are very few in other industries such as marine, tea, sugarcane, automobile, rubber, pulp and paper, sewage, and industrial waste, lignite/coal. The present review provides a thorough analysis of solar-based drying systems in agricultural and non-agricultural industries. Other industries can aware of the importance of solar drying so that they can also utilize this free energy. Therefore, the objective of this work is, (i) to review comprehensively the studies available on solar drying systems which are used in different industries, iii) to study different types of solar dryers, their design details and performance parameters involved, (iv) to get a collection of data in a single article by providing significant conclusions from the solar drying method used in different industries which may help the future researchers, (v) to review the economic, environmental, and social aspects of solar energy-based drying technologies and vi) to provide important information and points for future research in this specific area of solar drying.

Solar thermal energy for drying is an application of renewable energy that can be used by industries and also by farmers. Hence presented review paper helps the industries, farmers and researchers for the selection of appropriate dryers and it gives information on where solar thermal energy is used for drying, therefore, this topic is of potential interest to them. The article is structured as: Section 2 presents a brief report on the classification of solar drying systems based on their design, structure, type of material used for dryer, material to be dried, etc., Section 3 provides the different performance parameters of solar dryers. Section 4 gives information on different sectors where solar drying is employed. The economic, environmental, and social aspects of solar energy are discussed in Section 5. Section 6 provides few recommendations on drying systems and Section 7 furnishes the conclusions.

2. Classification of solar drying systems

A variety of dryers were noticed from different studies. Generally, these dryers are categorized based on their design, material used for the setup, type of assisted auxiliary heat source, the material used for the drying, material used for solar thermal energy storage (TES), etc. By considering these mentioned categories, a broad classification of dryers is drawn and shown in Fig. 1. All the solar dryer systems are explained very briefly in the following sections.

A. Lingayat et al.

ARTICLE IN PRESS



Fig. 1. Broad classification of various types of solar dryers.

2.1. Open sun drying (OSD)

The OSD is one of the commonly used methods of drying all over the world. In OSD, the material is spread over the ground or kept on the trays, floor and allowed to dry throughout the day by exposing it to open air and direct radiation of the sun (Muhlbauer, 1986). The risks involved in the OSD are; damage of material by birds, insects, and animals, quality degradation because of direct exposure of sun, dirt, dust, atmospheric pollution, quality degradation due to rain, air humidity dew, over-drying, insufficient drying, etc. (Arunsandeep et al., 2018). Due to these disadvantages involved in the OSD, people are searching for an alternate solar-based system that can partially control the input parameters so that a quality product can be improved.

2.2. Solar dryers

Some solar dryers maintain the controlled conditions required for drying such as temperature, velocity, humidity, etc. which are difficult in OSD. These controlled solar dryers are otherwise classified as indirect solar dryers (ISD) and direct solar dryers (DSD). It gives quality end products, low drying time, and dust-free products. These may be of natural convection type (passive method) (Lingavat et al., 2020a; Lingavat et al., 2017) or forced convection type (active method) (El-Sebaii and Shalaby, 2013). The overall efficiency of passive dryers varies from 20 to 40% which depends on the type of material, moisture content, temperature, velocity, and humidity of air (Udomkun et al., 2020). Most of the researchers are in favor of forced convection type dryers as they give better thermal efficiency and provide controlled drying conditions than natural convection dryers (Mustayen et al., 2014). Solar dryers can be provided with a TES system such as latent heat (Benli and Durmuş, 2009) or sensible heat storage (Ayyappan et al., 2016) unit to continue the drying during off sunshine hours.

2.2.1. Direct solar dryers (DSD)

The DSD is one of the commonly used alternate ways of OSD method. In this method, materials are kept in the chamber or in the space where they are exposed to the solar radiation transmitted through the transparent glazing. Transparent glazing can be any material, such as glass, plastic cover, polycarbonate material, etc. The DSD can be categorized as cabinet dryer and greenhouse dryer based on their design and construction and it can work in passive as well as active modes (Sontakke and Salve, 2015).

Cabinet dryer is a simple box type with transparent glazing for smallbatch drying as shown in Fig. 2. For mass drying, greenhouse dryers have been widely used. Various types of greenhouse dryers are reported in the literature and can be categorized based on the type of floor, type of glazing material, northern wall condition, dryer construction. The floor used in greenhouse dryers can be made of concrete and sometimes with a black coat. Polycarbonate, polyethylene materials, or PVC film-coated fiberglass are used for glazing material (Elhage et al., 2018). A greenhouse dryer can be a single or double sloped roof, dome-shaped or



Fig. 2. Cabinet or box type solar dryer.

A. Lingayat et al.

parabolic-shaped. The North wall provides in the greenhouse can be insulated or non-insulated. Singh et al., (2018) provided a detailed review of various types of greenhouse dryers and suggested few modifications. These dryers can provide a temperature in the range of 30 to 60 °C which is sufficient to dry fruits like apple, mango, papaya, strawberry, various types of vegetables, and other products like meat, marine products (Mezrhab et al., 2010; Rathore and Panwar, 2010; Singh et al., 2018).

2.2.2. Indirect solar dryers (ISD)

In this method, the product is not exposed directly to sun radiation but convective heat transfer between heated air and wet material is responsible for moisture removal (Lingayat et al., 2018; Lingayat et al., 2020c). Moisture removal rate and heat transfer can be controlled in a better way than DSD dryers. For improved airflow, natural convection systems can be replaced with forced convection systems (Mohanraj and Chandrasekar, 2009). Lingayat et al., (2020b) provided a detailed review of various types of ISDs and also suggested various possible locations for the TES system. A lot of researches have been reported to increase the thermal performance of ISD. Some of the advanced techniques can make the ISD more efficient and completely renewable by assisting with TES system with air heat exchanger, a solar water heater with water–air heat exchanger and all these will be discussed in Section 4.

2.2.3. Mixed-mode type solar dryer

Mixed-mode drying setups work on the joined principle of DSD and ISD methods. The dryer is provided with a transparent drying chamber and solar air heater for hot air generation. The drying time required to achieve the MC is less in mixed mode solar dryers compared with ISD because of the transparent drying chamber and solar air heater (Dhalsamant et al., 2018; ELkhadraoui et al., 2015).

2.2.4. Hybrid solar dryers (HSD)

In hybrid solar drying (HSD), auxiliary heating sources are used to dry the product during an off-sunshine hour and cloudy days or to maintain the constant temperature because of continuous fluctuation in solar radiation. Auxiliary heating sources such as an electric heater, biomass heater, liquid petroleum gas (LPG) heater, mechanical heat pump, etc. are used as a backup for thermal energy generation (Hossain et al., 2008; Mishra et al., 2020a; Yahya et al., 2017). This dryer is useful for drying high MC material as it protects the food from microbial attack because it is dried continuously (Mohanraj, 2014; Prasad et al., 2007; Thanaraj et al., 2007). The only disadvantage of the system is, it involves more manufacturing and maintenance costs and therefore it requires a specially designed structure.

Solar air collector (SAC) is the most important components in ISD, mixed-mode solar dryers and HSD. Some researchers reported various methods to increase the thermal performance of SAC and suggested different designs (Ozgen et al., 2009; Lingayat et al., 2020b). Esen et al. (2009a) suggested advanced modeling methods for its performance investigations such as artificial and wavelet neural network. The thermal storage system is also one of the important components in solar drying systems which stores the heat during peak sunshine hour which can be utilized after the sunset. Many researchers reported different methods to store the solar thermal energy in the form of sensible heat or latent heat (Esen, 2000; Esen and Ayhan, 1996; Esen and Durmus, 1998; Esen and Yuksel, 2013).

3. Various drying parameters involved in solar drying

Different types of solar dryers, as discussed in the previous section, have been used to generate the required air temperature for the drying. This section provides some basic theory involved during the drying process which is necessary to analyze the dryer performance and drying behavior of the material.

Solar Energy xxx (xxxx) xxx

3.1. Amount of water removed from the product and energy requirement

The mass of water removed from the material (M_w) during the drying process can be estimated as

$$M_w = m_i (MC_i - MC_f) / (100 - MC_f$$
(1)

Where m_i is the initial mass of material in kg, *MC* is the moisture content of the material on a wet basis. MC in the product can be estimated using, (Chandramohan, 2016a; Chandramohan and Talukdar, 2013).

MC, wetbasis(wb),
$$MC_i = (M_i - M_d)/M_i$$
 (2)

MC, drybasis(db),
$$MC_i = (M_i - M_d)/M_d$$
 (3)

Where M_i is the initial mass and M_d is the mass of the completely dried product in kg.

The initial MC of crops is generally estimated by an ASTM standard by heating them for 24 h at 105 $^{\circ}$ C in a hot air oven (Chandramohan and Talukdar, 2016).

The amount of heat required for the evaporation (E_{heat}) of moisture from the material can be calculated by the heat of evaporation by considering the humidity of drying air and can be mentioned as,

$$E_{heat} = \dot{m}_a (h_{do} - h_{atm}) \tag{4}$$

Where \dot{m}_a is the mass flow rate of air in kg/s, h_{do} and h_{atm} are specific humidity at the dryer outlet and ambient in kg/kg of dry air, respectively.

Also, the total energy needed to evaporate the moisture (Q_d) can be estimated by adding the sensible and latent heats (Ayensu, 1997),

$$Q_d = m_m C_{pm} (T_d - T_a) + m_w h_{fg}$$
⁽⁵⁾

Where, m_w and C_{pm} are mass and specific heat of the material in kJ/(kg K), T_a is surrounding temperature, T_d is drying temperature, h_{fg} represents the enthalpy of evaporation (kJ/kg) at T_d .

3.2. Efficiencies, save in drying time and specific energy consumption

The main function of solar dryers is to harness solar thermal energy using solar air collectors (SAC) or transparent cover sheets. The overall efficiency (η_0) of the collector can be calculated (Esen et al., 2009b),

$$\eta_o = m_{fluid} C_{pa} (T_c - T_i) / A_c I \tag{6}$$

Where, m_{fluid} is the mass flow rate of fluid in kg/s, T_i and T_c are temperatures of air at inlet and outlet of the collector, A_c is the area of the collector in m² and *I* is incident solar radiation (W/m²).

One of the performance parameters of the dryer is, save in drying time (t_s) which gives the drying time of ISD compared to OSD. It is calculated by,

$$t_s = \frac{t_{OSD} - t_{SD}}{t_{OSD}} \times 100\%$$
⁽⁷⁾

Where t_{osd} and t_{sd} are drying time in the open sun and solar dryer, respectively.

Drying efficiency (η_d) is the ratio of energy observed by the wet material for removing the moisture to the total energy incident on the collector. It can be estimated as,

Fornatural convection,
$$\eta_d = \frac{Q_d}{A_c I}$$
 (8)

Forforced convection,
$$\eta_d = \frac{Q_d}{A_c I + P_f}$$
 (9)

For the hybrid solar dryer,
$$\eta_d = \frac{Q_d}{\left(A_c I + P_f\right) + \left(m_{fuel} H_{fuel}\right)}$$
 (10)

A. Lingayat et al.

Where P_f is the fan power in kW, the product of m_{fuel} and H_{fuel} is the energy supplied by the auxiliary heating source.

Pick-up efficiency (η_p) indicates the moisture removal capacity of the drying air from the material and it is estimated as,

$$\eta_p = \frac{h_{do} - h_i}{h_{as} - h_i} \tag{11}$$

Where h_{do} , h_i and h_{as} are specific humidity of air at dryer outlet, inlet and saturated condition.

Specific energy consumption (*SEC*) (kJ/kg) is the ratio of total energy input (P_t) to the quantity of water removed from the wet product.

$$SEC = \frac{P_t}{m_w} \tag{12}$$

3.3. Exergy parameters

The maximum amount of available energy from the system is called exergy. The amount of exergy which is destroyed when entering into the dryer is called irreversibility (Esen et al., 2007b). It is useful to find the available energy at various regions within the dryer. The exergy values were estimated with the help of a generalized energy balance equation (Celma and Cuadros, 2009) and the final expression of exergy (E_x) was reduced after imposing assumptions (Tambunan et al., 2010) and is mentioned as,

$$Ex = \dot{m}_a C_{pa} \left[(T - T_{\infty}) - T_{\infty} ln \left(\frac{T}{T_{\infty}} \right) \right]$$
(13)

The exergy inflow (Ex_{dci}) and exergy outflow (Ex_{dco}) of the system are mentioned as,

$$Ex_{dci} = \dot{m}_a C_{pdai} (T_{dci} - T_{\infty}) - T_{\infty} ln \left(\frac{T_{dci}}{T_{\infty}}\right)$$
(14)

$$Ex_{dco} = \dot{m}_a C_{cpdo} (T_{dco} - T_{\infty}) - T_{\infty} ln \left(\frac{T_{dco}}{T_{\infty}}\right)$$
(15)

Where T_{dci} and T_{dco} are the temperatures at the inlet and outlet of the dryer.

The exergy input to SAC (Zhu et al., 2018) was estimated using:

$$Ex_{in} = \left[1 + \frac{1}{3}\left(\frac{T_a}{T_{sun}}\right)^4 - \left(\frac{4T_a}{3T_{sun}}\right)\right]IA$$
(16)

Where, T_{sun} is the apparent black body temperature of the sun, which is about 5770 K (Kalogirou et al., 2016)

The exergy losses are determined using exergy inflow (Ex_i) and exergy outflow (Ex_o) as mentioned below,

$$Ex_i = \sum Ex_i - \sum Ex_o \tag{17}$$

The exergy efficiency (η_{Ex}) is estimated by,

$$\eta_{Ex} = (Ex_i - Ex_l)/Ex_i = 1 - (Ex_l/Ex_i)$$
(18)

The improvement potential, IP is estimated as,

$$IP = (1 - \eta_{Ex})Ex_l \tag{19}$$

4. Solar dryers in different industries

According to International Energy Outlook (IEO), U. S. Energy Information Administration, (2019), 50% of the heat generated is consumed by industries, 46% of heat is consumed by space and water heating and only 4% of heat is consumed for agricultural greenhouse heating. Energy consumption in the sectors namely industrial, transportation, residential, and commercial increases day by day due to an increase in demand. It is supplied by commercial energy sources including renewable energy but its contribution is very small compared to other sources (Monthly Energy Review January 2020, U. S. Energy Information Administration, 2020). Among the 35–38% of total energy industrial consumers, 11.5% of industries need temperature less than 150 °C, 8.5% industries require a temperature between 150 °C and 400 °C, and remaining industries need more than 400 °C (Knaack et al., 2018; The International Energy Agency (IEA) (2018)). This pattern of energy consumption is almost similar in all industries all over the world (Mujumdar, 2006). The various industries and their thermal energy requirement for solar drying are discussed in this section.

4.1. Agricultural and food industry applications

Water content in the food material causes its spoilage due to microorganisms and bacteria growth. Drying restricts the growth and reproduction of microorganisms to avoid spoilage, and also it is used to reduce the weight and volume of the material which saves the transportation cost (Chandramohan, 2016b). Dehydration in agricultural products needs a large amount of heat which can be supplied by harnessing solar thermal energy. The utilization of solar energy also helps to reduce conventional energy sources. Many researchers provided a review on different types of dryers and carried out mathematical, analytical, and experimental studies for thermal analysis, and proposed different techniques for enhancing the dryer efficiency (El-sebaii and Shalaby, 2012; Lamidi et al., 2019). Various conventional solar air heaters and their design and performance parameters are reported in the literature (Ekechukwu and Norton, 1999; Lingayat et al., 2020c; Mohana et al., 2020). A detailed review of the effect of various influencing parameters has been discussed by Mishra et al. (2020b).

Lakshmi et al., (2018) found the drying kinetics of black turmeric using a mixed-mode forced convection solar dryer integrated with a TES system. The moisture removal rate of turmeric has been examined by comparing the results with OSD. Two sets of thin layered black turmeric samples (each 200 g) were chosen for the investigation. The first set was placed in a solar dryer and another was placed for OSD. A shell and tube heat exchanger TES system was coupled with the solar dryer to dry the samples during night times. 35 kg of paraffin wax is used as an energy storage material. It was observed that the discharge air temperature at the outlet of the system was 6 – 10 $^\circ C$ more than the ambient temperature at night times. Turmeric was dried from an initial MC of 73.4% to 8.5% (wb) using a solar dryer and OSD which took 18.50 h and 46.50 h, respectively. The specific energy consumption of the product to dry was estimated and is 5.21 kWh/kg per kg of moisture. The average collector efficiency and overall drying efficiencies were found to be 25.6% and 12%, respectively. The quality of fresh and dried turmeric was tested and found better results for solar dried turmeric in terms of color, phenolic content, and flavonoid content. Similarly, there were other studies found with solar dryers, and the products dried are; red chili (Ndukwu et al., 2017; Rabha et al., 2017), bitter gourd (Zachariah et al., 2020), garlic (Shringi et al., 2014), apricot (Baniasadi et al., 2017), apple (Atalay et al., 2017) and orange (Atalay, 2019). Each studies' data such as drying capacity, final mass, initial and final MCs, average ambient and drying air temperatures, drying time, energy storage material, total and specific energy consumptions, the mass of water evaporation, dryer, collector, and exergy efficiencies are mentioned in Table 2.

It is observed from Table 2 that the highest drying air temperature (65 °C) was noticed at Lakshmi et al., (2018) among all other studies. The overall drying efficiency (η_d) is just 12% as mentioned in Table 2 because of plastic sheet is used as a collector cover instead of antirefluxing glazed glass. The transmissivity of glazed glass is 0.9 therefore it can transmit 90% of the total incident solar radiation. Also, the glazed glass can transmit the solar heat flux in one way only. It cannot allow the reflected solar radiation from the absorber plate to the outside atmosphere. There is a possibility of improvement in collector design as the setup produced a low drying efficiency (12%) because of the huge

A. Lingayat et al.

Table 2

Selected studies on solar drying of agricultural crop products and their drying and performance parameters.

Parameter	Lakshmi et al., (2018)	Ndukwu et al., (2017)	Ndukwu et al., (2017)	Zachariah et al., (2020)	Shringi et al., (2014)	Baniasadi et al., (2017)	Atalay, (2019)	Mohanraj and Chandrasekar, (2009)	Rabha et al., (2017)
Dryer Size/ Specifications	2.04 m length, 1 0.04 m width and 0.2 m depth	-	-	$\begin{array}{l} 280 \text{ mm} \times \\ 120 \text{ mm} \times 80 \\ \text{mm} \end{array}$	1.2 m × 0.6 m × 0.5 m	$\begin{array}{c} 0.7 \text{ m} \times 0.6 \\ \text{m} \times 0.15 \text{ m} \end{array}$	2.3 m × 2.3 m × 2.3 m	$\begin{array}{c} 1 \text{ m} \times 1 \text{ m} \times 1.5 \\ \text{m} \end{array}$	$\begin{array}{c} 2 \text{ m} \times \\ 0.85 \text{ m} \times \\ 0.37 \text{ m} \end{array}$
Product dried	black turmeric	Red chili	Red chili	Bitter gourd	Garlic clove	apricot slices	Orange slices	Chili	Ghost chili pepper
Drying capacity (kg)	15	1	1	2.5	_	_	10	40	9
Final mass of the product (kg)	-	0.373 kg	0.398 kg	-	-	-	0.988 kg	-	1.02 kg
Initial moisture content of product (%)	73.4%	72.2%	72.2%	92%	55.5%	86%	93.5%	72.8%	85.5%
Final moisture content of product	8.5%	7.6%	10.1%	4%	6.5%	25%	10.76%	9.1%	9.7%
Average ambient temperature (°C)	-	28.25 °C	28.6 °C	32 °C	32 °C	-	27.3 °C	31 °C	32 °C
Drying air temperature (°C)	65 °C	40.35 °C	41.83 °C	48 °C	39–69	65 °C	55.4 °C	50.4 °C	50 °C
Drying time (h)	18.5 h (solar drying) and 46.5 h (open sun drying)	24.5 h	36.5 h	10 h	8 h	1.55 gr/h	7.2 h	24 h	42 h
Energy storage material	Paraffin wax	sodium sulfate decahydrate (Na₂SO₄·10H₂O)	Sodium chloride (NaCl)	PVC caps, Al pipes and paraffin wax	PCM	Paraffin wax	Packed bed (Pebbles)	Gravel	Heat storage module
Total energy consumption (MJ)	-	7.37 MJ of solar energy and 0.204 MJ of thermo-chemical energy	11.49 MJ and 0.011 MJ of solar and thermo- chemical energy	_	-	-	88.1 MJ	_	21.3 MJ
Specific energy consumption (kWh/kg per kg of moisture)	5.21	3.34	5.28	6.1 (without PCM) and 3.6 (with PCM)	-	-	1.98	-	18.72
Mass of water evaporate (kg of water / kWh)	0.192	0.627	0.602	0.16 (without PCM) and 0.28 (with PCM)	0.54–1.05	0.933	0.505	0.87	0.053
Relative humidity (%)	22% (inside) and 65% (outside)	42.23% (inside) and 64.9% (outside)	59% (inside) and 64.9% (outside)	-	-	-	45.21%	60%	-
Drying efficiency (%)	12	18.79	11.89	18.6	_	10.7	34.4	21	4.05
Collector efficiency (%)	25.6	-	-	-	_	_	-	-	22.95
Exergy efficiency (%)	-	66.82	-	67 – 88	-	-	63.34	-	63

amount of heat losses such as convection and radiation.

A newly developed mixed-mode solar dryer is a better option for drying high-value medicinal agricultural products like black turmeric (Lakshmi et al., 2018). In DSD dryers, the food product directly absorbs the solar radiation through the collector cover, whereas, in ISD, the air is heated in a separate chamber and sent to the drying chamber. The mixed-mode dryers have the provision of both DSD and ISD dryers, therefore, it has the following advantages;

- A rapid drying rate is possible with the safe moisture level in the product.
- The time required to dry the products is less compared to DSD and ISD drying techniques.
- Uniform distribution of airflow rate is possible inside the dryer.

A natural convection solar dryer (NCSD) was integrated with TES materials such as sodium chloride (NaCl) and sodium sulfate decahydrate (Na₂SO₄·10H₂O) for continuous drying of red chili at Nigerian climate conditions (Ndukwu et al., 2017). Three cases of experiments were conducted namely; dryer with Na₂SO₄·10H₂O (Case-I) and NaCl (Case-II) TES systems and dryer without TES system (Case-III) was used to dry red chili to reduce the MC from 72.27% (wb) to 7.6% (wb), 10.1% and 10.3%, respectively. It took a drying time of 24.5 h, 36.5 h, and 40.5 h for case-I, II, and III, respectively. There is not much considerable difference in drying time of Case-II and Case-III experiments. But the considerable difference is there in Case-I and Case-III and almost 16 h drying time is saved in Case-I. Therefore, Na₂SO₄·10H₂O can be used as an efficient TES material (Ndukwu et al., 2017).

The overall drying efficiency and energy utilization of the three cases

A. Lingayat et al.

differed from 10.61 to 18.79% and 7.54–12.98 MJ, respectively. The exergy efficiency of the dryer at night times (with NaCl and Na₂SO₄·10H₂O) was 81.19% and the same for the overall drying process (round-a-clock) was 66.82%. It was concluded that approximately 602 tonnes/year of CO₂ could be limited from entering the air using Na₂SO₄·10H₂O as a TES medium compared to a diesel-powered dryer. The most important finding is effective energy utilization. Case-I, II, and III have consumed energy of 3.34, 5.28, and 5.92 kWh/kg of moisture removal. It proves that these types of NCSD integrated with TES effectively minimized various heat losses which lead to efficient energy usage.

A novel photovoltaic (PV) assisted greenhouse cabinet mixed-mode solar dryer integrated with a TES system was used for drying bitter guard slices (Zachariah et al., 2020). The paraffin wax was used as TES material which was packed into the dryer cabinet. 2.5 kg of the bitter guard were dried from 92% to 4% MC (wb) which took 10 h with the TES system and the same took 1 day without the TES system as mentioned in Table 2. The required specific energy per kg of moisture removal from the product was 3.6 kWh per kg of water (with TES system) and 6.1 kWh per kg of water (without TES system), respectively. The specific moisture removal rates from the product (bitter gourd) with and without the TES system were 0.28 and 0.16 kg of water per kWh, respectively. The overall η_d of the solar dryer with and without the TES system was 18.6% and 10.8%, respectively. The primary findings are; an increase in overall η_d and a reduction in drying time. It just took 10 h to completely dry the product which is a good sign in the development of mixed-mode type solar dryers. A similar kind of study was carried out by Shringi et al., (2014) for drying of garlic clove. The cloves were dried from 55.5% to 6.5% MC (wb) at the drying time of 8 h. The energy and exergy efficiencies were between 67.06% and 88.24 % and 3.98% to 14.95%, respectively.

An experimental investigation of a mixed-mode forced convection solar dryer (FCSD) integrated with a TES system was carried out by Baniasadi et al., (2017). A blower was used to maintain a continuous supply of air for drying apricot slices. Paraffin wax was used as a TES material and it was placed at the bottom of the cabinet tray. The basic difference between this system and other mixed-mode dryers was the structure and construction of the solar collector. The collector plate was tilted to 30° and the total collector area was partitioned into three zones in the form of trapezoidal shapes. Experiments were carried out with and without the TES system to remove MC from initial MC of 86% to final MC of 25% as shown in Table 2. The dryer with a TES system provides a 50% higher drying rate than without a TES system. The η_d and moisture removal pickup efficiencies were 10.7% and 9.8%, respectively.

The impact of packed bed TES system and waste heat recovery system (recuperator) was examined on the performance of a solar dryer which was used to dry apple slices (Atalay et al., (2017). The novelty of the dryer was the drying air was again and again recirculated with the help of a recuperator. So that 50 to 60% of the waste heat was utilized effectively in the dryer. The average temperature of the air was between 55 and 60 °C as presented in Table 2. The stored thermal energy in the packed bed system was 210 MJ. The drying process was continued till the final moisture of the apple slices was reached 10%. The average drying time was 6 h. The drying process took 8 kW of energy to complete the drying process. The specific moisture removal rate of the product was 0.819 kg per kWh.

The practical feasibility of a low-cost solar drying system integrated with a packed bed TES system was investigated using an experimental study (Atalay, 2019). The drying kinetics of orange slices were studied to investigate various performance parameters such as exergy, energy, sustainability index, improvement potential, and waste exergy ratio. Orange slices were dried from 93.5% to 10.28% MC (refer to Table 2). The average energy required and specific energy consumed to dry the orange slices was 89.89 MJ and 1.89 kWh per kg, respectively. The specific water evaporation rate (SMER) was 0.53 kg per kWh. The exergy efficiency was in the range of 50.18–66.58 % and 54.71–68.37% for

dryers without and with the TES system. The sustainability index and improvement potential were in the range of 2.15 to 3.17 and 0.016 to 0.07 kW, respectively. This system has improved drying efficiency (34%) and SMER because of minimum heat losses from the system.

An experimental analysis was performed in an indirect FCSD with TES material (gravel) to dry red chili (Mohanraj and Chandrasekar, 2009). The MC of the red chili was reduced from 72.8% to 9.2%. The average drying air temperature inside the cabin was 50.4 °C (refer to Table 2). A high drying rate of 6.2 kg/kg of dry matter per hour was achieved in the initial stages of drying. The thermal efficiency of the solar dryer was 21% with a specific moisture removal rate of 0.87 kg/kW h. A similar analysis with energy and exergy analysis in an FCSD was carried out by Rabha et al., (2017). A paraffin wax-based shell and tube latent heat storage unit was used in the setup. It was examined with 20 kg of red chili with an initial MC of 73.5% (wb). The product was dried up to the MC of 9.7% (wb) in four consecutive days (the effective drying period is 35 h). In comparison with the OSD method, the newly developed FCSD method saved 122.8% of the drying time of chili. The exergy and energy efficiencies of the TES unit were between 18.3 and 20.5% and 43.6-49.8%, respectively. The exergy efficiency of the drying cabinet was in the range of 24.6% to 98.1% with an average of 52.2%. The specific energy requirement and the gross efficiency of the solar drying system were 6.8 kWh per kg of moisture and 10.8%, respectively. The power consumption of the dryer was 0.7 kWh per kg of moisture which was only 10.3% of the specific energy consumption of the chili.

A similar study of FCSD with paraffin wax was performed by El Khadraoui et al., (2017). The performance was evaluated at two different cases under no-load conditions: with PCM and without PCM. The daily energy and exergy efficiencies of FCSD with PCM were 33.9% and 8.5%, respectively. At night time, the drying air temperature inside the drying cabinet was 4 to 16 $^{\circ}$ C higher than outside ambient temperature. A similar kind of research was already attempted by El-Sebaii and Shalaby, (2013) for drying thymus and mint under Tanta, Egypt weather conditions.

An HSD dryer for osmotically pre-dehydrated cherry tomatoes was designed by Nabnean et al., (2016). 100 kg cherry tomatoes were dried at a temperature between 30 and 65 °C for 4 days. The osmotic pretreatment of tomatoes before solar drying helps to improve final quality. Sekyere et al. (2016b) developed a solar dryer with six infrared heat lamps to simulate solar radiation in laboratory conditions. Concrete absorber and rock pebble bed were used to store solar energy and assisted with an electric heater. Performance of dryer was investigated with 2.3 kg pineapple slices and the MC in the solar dryer was reduced from 912 to 155% (db) in 7 h. Yahya et al., (2018) carried out the performance and economic analyses on solar-assisted heat pump fluid-ized bed dryer integrated with biomass furnace for rice drying. MC of rice was reduced from 32.85% (db) to 16.29% (db) in 22.95 min, with a mass flow rate of 0.1037 kg/s at an average temperature of 80.9 °C. The payback period of the system was1.6 years.

In recent days, new parameters such as waste exergy ratio, sustainability index, and improvement potential are introduced to assess the performance of dryers with storage systems (Atalay, 2019). From the recent studies related to the TES device, the drying efficiency was 21% (Mohanraj and Chandrasekar, 2009) which was less than 20% a decade before. It was proved that FCSD with a TES storage system can enhance the performance and overall efficiency of the drying system. One more new term (pick-up efficiency) is introduced by Baniasadi et al., (2017) to evaluate the actual performance of the solar dryer based on the removed moisture from the food product. The structure of a dryer suggested by Baniasadi et al., (2017) is relatively less complicated and never utilized the auxiliary energy during daytime. The drying rate was reduced by 50% with the use of energy storage material. The dryer integrated with the TES system could recover an energy payback period of 1.91 years and also a discounted payback time of 0.8 years which is lower than the life span of the dryer making it environmentally and economically safe (Zachariah et al., 2020).

A. Lingayat et al.

4.2. Marine industries

Marine industries also one of the major sectors where solar energy plays an important for the drying of married products. A solar dryer with a TES system was developed for shrimps drying (Murali et al., 2020). The water-based TES system stored the energy during peak sunshine hours. Also, the dryer was assisted with liquefied petroleum gas (LPG) as an auxiliary heating source. 50 kg shrimp were dried from 76.71% (wb) to 15.38% (wb) within 6 h of drying as mentioned in Table 3. Solar collector supplied 73.93% of heat and the rest of the energy (26.07%) was assisted by the LPG water heater system (Murali et al., 2020). Table 3 provides the various studies on the solar drying of marine applications.

An indirect type hybrid FCSD dryer with an artificial heat source of diesel burner was designed and developed by Fudholi and Sopian, (2019) to dry salted silver jewfish. V-groove collector was used to enhancing the heat flux absorbing capacity of the collector. The diesel burner was assisted with the dryer so that the drying could be effectively performed with a continuous supply of heat. The dryer took a minimum of 8 h to remove the MC from 64% to 10% (wb). The specific energy consumption rate of the dryer was 2.92 kWh/kg (as shown in Table 3). The drying and exergy efficiencies of the hybrid system were 23% and 31%, respectively.

The same marine food (silver jewfish) was dried in another study (Bala and Mondol, 2001). They investigated the performance of a solar tunnel dryer at Bangladesh atmospheric conditions. A 20 m length and 2 m width of tunnel dryer were used for the drying process. MC was reduced from 67% (wb) to 16.78% (wb) in 36 h. This chemical composition of dried fish indicated that the fish dried in the solar tunnel dryer was a quality product for consumption. A hybrid solar dryer assisted with biomass-based air heating system had been tested by Hamdani et al., (2018) for fish drying. Solar energy was used from 9.00 am to 4.00 pm followed by a biomass burner from 4.00 pm to the next day at 6.00 am. Overall, the drying air temperature was maintained at 40–67 $^\circ\text{C}$ and the mass of fish reduced to 12.5 kg from 25 kg (as mentioned in Table 3). The maximum drying air temperature maintained inside the cabin was 50 °C (in the daytime) and 67 °C (at the night with the help of a biomass dryer). The total moisture removal rate from the product (on a wb) is 50.5% to 12% within a drying period of 23 h. The total incurred cost of the hybrid solar dryer was \$ 1870 with the break-even point of 2.6 years.

The technical feasibility of industrial solar dryer integrated with linear Fresnel collector and concrete TES for production of pasta was conducted by Liu et al., (2015). A dryer capacity of 2.4 MW was used. Thermic oil was heated with a linear Fresnel collector. Hot thermic oil was used to heat the water with the temperature range of 120 to 140 $^{\circ}$ C

Table 3

Selected studies on solar drying of marine applications

which was then supplied to heat the air for pasta drying. The proposed solar dryer could save 40% annual thermal energy in a pasta factory.

Marine foods are highly sensitive tissues that need to be dried with much care. The partial moisture within the food destroys the tissues and it may be unusable then. The safer MC within the fish is reported as 15–20% (wb) (Rahman, 2006). The studies with 15–20% of final MC were not found for fish drying in literature. Such an intensive drying process, the dryer needs a lot of energy and it cannot be feasible by solar energy alone or need more researches to achieve the same. In such a situation, hybrid dryers can be designed (Hamdani et al., 2018) to run the dryer round-a-clock so that the micro-organisms reaction can be avoided. The possible options are solar energy with bio-mass energy (Hamdani et al., 2018) or industrial waste heat energy, or diesel burner (Fudholi and Sopian, 2019).

Most of the time, the marine foods were dried with dry salts. The dry salts always take further MC from the tissues. When the muscles were with more MC and dried with a higher temperature, the salt is diffused uniformly all its geometry because the diffusion coefficient is a function of temperature and MC. A good salting process was explained for Tilapia fish in (Rahman, 2006). The fish-salt ratio is 3:1, salting time was up to 24 h, the effective drying time of 6–20 h with the drying air temperature range of 40 to 60 °C are the other drying data provided in (Rahman, 2006). From the experimental works of marine food drying, it was found that the average collector exit temperature was 75 °C with an average drying efficiency of 25.42%. The average temperature attained inside the chamber lies between 60 and 65 °C.

4.3. Tea industries

Tea is one of the most consumed drinks. China and India have a major contribution to tea production which is near about 29.56% and 25.35% of the total world tea production (Chang, 2015). Withering, drying, grading, and packing are processes involved in tea production that are energy-intensive and these processes consume 85% of thermal energy and 15% of electrical energy (Sharma et al., 2019). The thermal energy required for the drying process during tea production can be fulfilled by solar energy. Vacuum-assisted solar dryers were designed and experimented for drying black tea by Pou and Tripathy (2020). The relationship between input parameters such as vacuum level, loading rate, and output parameters such as color, aroma, index, drying time and energy consumption were presented. The system was optimized based on the maximum liquor color and aroma index with minimum energy consumption and drying time. Liquor color, aroma index, drying time and energy consumption were estimated to be 20.08, 11.65, 4.44, 140.66 min and 21450.7 kJ, respectively, at an optimum vacuum level of 570.71 mmHg and loading rate of 0.96 kg m⁻². Ozturk and Dincer

Paper	Size / Specifications	Product dried	Drying capacity (kg)	Initial moisture content	Final moisture content	Air Temperature (°C)	Drying time (h)	Heat source	Energy consumption (kJ)	Drying efficiency (%)	Exergy efficiency (%)
Murali et al., (2020)	$\begin{array}{l} 4.23 \text{ m} \times 1.00 \\ \text{m} \times 1.80 \text{ m} \end{array}$	shrimp	50	76.71% (wb)	5.38% (wb)	55	6	Solar system (2,06,641 kJ) and LPG water heater (72,890 kJ)	2,79,531	37.09	_
Bala and Mondol, (2001)	40 m ² area	Salted silver jewfish	150	67% (wb)	16.78% (wb)	52.2	36	Solar energy	-	_	-
Fudholi and Sopian, (2019)	13.8 m ² area	Salted silver jewfish	51	64% (wb)	10% (wb)	50	8	Solar system and Diesel burner	2.92 kWh/kg	23	31% (Avg.)
Hamdani et al., (2018)	$2.6 \text{ m} \times 0.8 \text{ m}$	Queen fish	25	50.5% (wb)	12% (wb)	50 (day time), 67 (night)	23	Solar system and biomass fueled air heater	_	_	-

A. Lingayat et al.

(2019) conducted the exergo-economic analysis of solar dryer integrated with PV unit for drying of tea leaves. MC in 100 kg wet tea leaves was reduced from 80% (wb) to 3% (wb) in four days. The capital cost and the capital productivity of the dryer were estimated to be \$5953 and 1.54, respectively. The energetic loss ratios, exergetic loss ratios, exergy efficiency and exergy destruction were estimated as 76.45 MJ/\$, 72.63 MJ/\$, 74% and 201.6 GJ, respectively.

Chamomile leaves (herbal tea) were dried in a solar dryer and the drying kinetics were estimated by Amer et al. (2018). The dryer was assisted with a solar collector, reflector, water air heat exchanger, and solar heated water storage tank with an electric heater as a supplementary heat source. Water was used as a sensible heat storage medium which stored heat energy during day time. Two drying chambers were used with a capacity of 32–35 kg. One drying chamber was placed just below the absorber surface and the second one was placed in a series with the first chamber. Temperature controllers were used to controlling the required drying temperature. The system was useful to dry chamomile from 75% to 6% (wb MC) which helped to reduce the drying time by 27–30 h compared with OSD. Midilli model was suggested for fitting the drying kinetics of chamomile.

An FCSD dryer with an auxiliary heating system was used to dry Mexican tea leaves (Kane et al., 2008). During the experiments, the following parameters such as solar insolation, ambient temperature, drying air temperature, relative humidity, and airflow rate were varied. It was concluded that the temperature was the most influential parameter during drying. The diffusion coefficient of tea leaves was in the range of 1.0209 to 1.0440×10^{-9} m².s⁻¹ and the activation energy was 89.15 kJ.mol⁻¹. They reported that Wang and Singh model is the best model for the characterization of Mexican tea leaves drying.

The economic viability of roof-integrated solar dryer for tea drying was investigated by Palaniappan and Subramanian (1998). The dryer was provided with other sources of heat energy such as coal and firewood. The built collector area was 212 m^2 . The solar energy usage on the dryer saved the annual fuel consumption by 25%. Sopian et al. (2000) carried out the experimental analysis on a solar-assisted drying system for herbal tea drying. The solar collector was provided with V-grooves so that it could create enough turbulence and hence enhanced the heat transfer rate. The collector area of 20 m^2 could provide a hot air temperature of more than 50 °C at a flow rate of 15.1 m³/min. An auxiliary heating system was employed if the air temperature was less than 50 °C. The dryer effectively drying the herbal leaves from 87% MC (wb) to 54% MC (wb) in 12 h.

From the above studies on tea leaves drying, the following highlights were observed. Vacuum-assisted solar dryers help to produce quality black tea after providing the optimum vacuum level and loading rate (Pou and Tripathy, 2020). Exergo-economic analysis needs to be performed to find the energy and exergy losses from the dryer. Also, expenses incurred for the drying system and the payback period can be identified as these are important terminologies for tea industries. A hybrid solar dryer with a reflector, thermal storage, and two drying chamber could effectively help to save the drying time up to 20 h compared to OSD. These types of dryers can be commercialized in rural areas so that small-scale farmers get benefitted (Amer et al., 2018). The annual fuel consumption is reduced up to 26% by pre-heating of teaprocessing air using a solar air heating system (Palaniappan and Subramanian, 1998).

4.4. Automobile industry

The automobile industry has a long and complex manufacturing process. Paint curing is an important process in automobile industries which is carried out by thermal energy mainly in the form of hot air at a temperature in the range of 80–150 °C. Also, during the manufacturing process of automobile components, hot air is used to dry the components at a temperature below 90 °C. This heat energy can be retrieved from solar energy instead of using a commercial energy source. The

automobile sector is one of the developing sectors as well as expensive, the usage of solar energy is highly advantageous. Also, it has a lot of potential for the involvement of a solar air heating system for various operations. SAC or water-based collector with a water-air heat exchanger can be used for air heating up to 80 °C to 150 °C which can save a lot of fossil fuel consumption (Giampieri et al., 2020; Orsato and Wells, 2007).

The polymeric coating is another important application in the automobile sector. It is used to avoid corrosion in steel plates. It is also advantageous such as less toxic and environmentally friendly compared to other chemical coatings (Singhal et al., 2014). It is made up of layer by layer and each layer needs to be properly dried to make the subsequent layer. A high temperature drying air up to 90 °C is necessary to dry each layer. Solar drying can be preferred by providing proper arrangements so that the excess cost involved in commercial energy can be avoided. More understanding is required between chemists and drying researchers to take forward this specific research area with solar drying to the next arena.

4.5. Rubber industry

Rubber is an elastic substance used in a wide range of goods production. Rubber is produced by natural latex from the plant or by artificial synthetic method. Drying plays an important role in maintaining the quality of the final rubber sheet from raw material. Breymayer et al. (1993) reported that the improper drying of raw rubber material results in low-grade rubber sheets which are about 80% of the total production which have to be sold at relatively low prices. Drying is generally done by a simple hot air-drying method or by smoke drying method. In the airdrying method, wet rubber sheets are dried with the OSD or by producing the hot air by conventional energy sources. In the case of the smoke-drying method, smoke was generated by burning firewood and biomass in the smokehouse to remove the moisture from rubber sheets. Rubber is generally dried in the temperature range of 45-60 °C and it can be easily achievable using solar energy (Tanwanichkul et al., 2013). Breymayer et al. (1993) developed the dryer integrated with SAC and traditional firewood/biomass smokehouse as an auxiliary heating source. SAC with recirculation of exit air from the dryer helped to generate a drying air temperature range of 45-60 °C. A total of 320 kg of rubber sheets were dried using a solar dryer and the sheets lost their MC from 60% to 0.5% which reduced the firewood consumption. It took five days to achieve 0.5% MC in the solar dryer with biomass whereas, it took seven days with the solar dryer alone. But the developed system is very advantageous because if commercial energy is used for seven continuous days (instead of solar drying), the expenses will increase further. An ISD dryer assisted with an electric air heater was used for natural rubber drying which provided the quality rubber as a final product (Pratoto et al., 1998). The moisture content of rubber sheets was dried from 60% to 0.5% (wb) in 8.8 h at 100 °C.

Dejchanchaiwong et al., (2016) compared the drying performance of mixed-mode and ISD dryers while drying with 30 natural rubber sheets. Better quality and drying rate was noticed for the mixed-mode dryer compared to the ISD. Both dryers helped to reduce the drying time by 2–3 days compared to OSD. Dryer efficiencies of the mixed-mode and ISD dryers were 15.4% and 13.3%, respectively. MCs in rubber sheets were reduced from 32.3 to 2% (wb) and 29.4 to 8% (wb) in four days using mixed-mode and ISD dryers, respectively.

The performance of the greenhouse solar dryer for rubber sheet drying was analyzed by Janjai et al. (2015). The dryer was able to provide the temperature from 32 to 55 °C which was useful for effective drying of 750 kg rubber sheets without affecting its quality compared with OSD. The dryer was useful to reduce the MC of rubber sheets from 24 - 30% (wb) to 0.4–3.9% (wb) in five days where OSD could reduce only 13% in five days. They also developed an ANN model to predict the drying characteristics of rubber sheets.

Artificial neural network modeling was carried out to develop the

A. Lingayat et al.

dryer for mass production. Jitjack et al. (2016) suggested a parabolic greenhouse dryer with area-enhanced panels for effective drying of rubber sheets as shown in Fig. 3. There were two setups made to compare the results. One was made with a parabolic greenhouse collector and in another, additional collector plates were provided for enhancing the collecting area. Better quality of rubber sheets was noticed in the enhanced area model compared with simple parabolic greenhouse dryer. The dryer was able to provide a maximum temperature of 55 °C, which helped to reduce the drying time by 3.5 days compared to the dryer without area-enhanced panels.

When the rubber sheets are dried with natural and OSD drying methods, the final dried product took a lot of drying times, and sometimes it took 10-20 days. Such a long drying time creates huge color changes on the final product. Sometimes, more MC and long drying time produce fungus growth on the product and thereby deteriorating the natural color further. These are all not encouraged by the purchaser or the product price is reduced from the standard price which creates a lot of inconvenience to the rubber farmers. Therefore, the rubber sheets need to be dried within few days so that the freshness is maintained which attracts the buyer. Such issues can be solved using solar drying techniques. A perfectly designed solar drver produces up to 70 °C (Lingavat et al., 2020a) air temperature inside the drying chamber, therefore, rubber sheets can be dried enough in these setups. More experimental and numerical studies are required to run the setup with different volumes of rubber sheets so that industries come forward to make their setup and finally it can reach farmers.

4.6. Pulp, paper, and allied industries

The pulp and paper industry is one more sector where solar-based technologies can play an important role to reduce energy costs. Different processes involved in these industries are pulp making, bleaching, paper making, drying, etc. Drying is the process that happened at the final stage at a temperature between 60 °C and 150 °C. It is a very important and critical process that consumes a large amount of heat (Gemechu et al., 2012). The commonly used method of OSD for handmade paper can be replaced by solar drying (Madhavan and Ramachandran, 2015). An ISD dryer was developed for drying twelve A4 size paper sheets (each 22 g) at a time. The collector outlet temperature was in the range of 45 to 70 °C. The dryer was useful to reduce the MC of sheets from 50% to 7% (wb) during 65 to 75 min of drying time. An economic analysis was carried out and found that a payback period for the dryer was 95 days.

Hjort and Thomas, (2014) developed a small prototype of a solar dryer assisted with a water heater and air–water heat exchanger for the



Fig. 3. The pictorial view of greenhouse solar dryer with natural rubber sheets inside (Jitjack et al., 2016).

application of paper drying. The prototype was tested for different handmade paper types at different input conditions. The specific energy consumption was 3.5 MJ / kg of water evaporated and the dryer efficiency was 80%. The developed prototype is useful for developing a large-scale setup.

The potential of solar energy for process heat and its advantages on the reduction of CO₂ emissions in the paper and pulp industry was presented by Sharma et al. (2016). Annual process heat requirements were estimated based on location and raw material used and estimated the performance of commercial systems. Tagnamas et al. (2021) used a convective solar dryer to find the drying kinetics and characteristics of the carob pulp at various drying temperatures from 50 to 80 °C and velocity of 0.18 and 0.09 m/s. It was observed that the diffusion coefficient varies with temperature. It was in the range of 1.56 to 6.98×10^{-9} m²/s. The energy and exergy efficiencies were estimated and these are in the range of 4.23% to 7.25% and 30.12% to 80.5%, respectively.

Among the above studies, it is identified that the pulp and paper industry sector also reliably used solar energy. There were many setups developed for the paper and pulp drying process. This sector needs to be addressed well for future development. It needs more studies so that these industries can develop their system to dry the materials with solar energy. The thermal energy and the quality of energy needed for pulp and paper industries need to be estimated. The other major thermal energy requirement factors such as land availability, location, fuel cost, quality of steam required, compatibility of existing method, other emergency heating systems, etc. need to be investigated for the benefit of this sector.

4.7. Sugarcane industry

Bagasse is crushed fibers left over after the juice from the sugarcane industry which is an excellent raw material that can be used as fuel for electricity generation, bio-methane production, animal feed, and fertilizers (Rabelo et al., 2011). Generally, the dry bagasse is used as fuel to heat the cane juice in the heating pan. For effective combustion, it is necessary to remove the moisture from the bagasse (Venkata Sai and Reddy, 2020). Few studies reported that the solar drying method can be chosen for drying the wet bagasse before feeding into the furnace for combustion (Phadkari et al., 2017; Subahana and Natarajan, 2016). Phadkari et al. (2017) developed an FCSD setup for solar drying of bagasse pulp where 1000 g of bagasse fibers were dried. The proposed dryer was useful to dry the bagasse from the initial MC of 48% to 16.77% (wb).

Theoretical and experimental analysis of solar tunnel dryer was carried out by Subahana and Natarajan, (2016) for drying of switchgrass, rice straw, and sugarcane trash. The setup was made with four compartments and the bottom of the compartment was made with wood for avoiding heat losses. Aluminium chambers with stainless steel meshes were provided for keeping the materials. Four different soils such as red (absorptivity of 0.65), yellow (0.73), grey (0.84), and black (0.95) were used in four compartments to check the effect of absorptivity of soils. The rice straw was dried for 8 h and its MC reached 0.906, 0.872, 0.849, and 0.837 kg/kg of db for red, yellow, grey, and black soils, respectively. It shows that when the absorptivity of soil is increased, the moisture removal rate is decreased. When absorptivity increases, more portion of heat energy is absorbed by the soil results in lower temperature inside the compartments. It creates a low drying rate of rice straw. The setup was given with a provision to conduct the experiments for parallel and cross-flow passage. The crossed flow gave much better results because of more turbulence inside the compartments and hence the grasses get high exposure to hot air. Maximum temperatures of 78 and 63 °C were observed during theoretical and experimental analysis of rice straw drying with yellow-colored soil.

Vijayaraj and Saravanan (2008) carried out a numerical study on the drying behavior of rectangular bagasse to predict the temperature and moisture distribution within it. Variation in mass transfer coefficient and

A. Lingayat et al.

heat transfer coefficients were estimated which varied in the range about 0.0001 to 0.0125 m/s and 0.01 to 0.5 W/m²K, respectively. Embong et al. (2016) proposed a method to extract the SiO₂ from sugarcane bagasse ash. The authors suggested a pre-treatment method using low-concentration acid followed by a solar drying process. The proposed method was efficient for the production of sugarcane bagasse ash with high pozzolanic reactivity to extract SiO₂ that can be used as an alternative to cement. The solar dryer was used for drying sugarcane bagasse after washing in a constantly rotating cylinder for 20 min.

Sugarcane bagasse is an effective feedstock in combustion chambers and is also used for bioethanol production. This waste consists of a huge of MC and it needs to be removed properly. This moisture removal process can be done with solar drying. OSD drying creates high contamination on the final dried product which may affect the combustion process and also more studies need to be made with ISD dryers. The energy, exergy, and economic parameters need to be estimated and analyzed thoroughly so that more industries can utilize this concept which may directly benefit the farmers.

4.8. Sewage and industrial waste

Sludge and wastewater are a mixture of liquid and solid-like municipal waste or industrial waste that is discharged into the river or sea without pretreatment. A significant quantity of waste, in the form of sludge, is produced after wastewater treatment which carries 95% of water content. Presently, mechanical systems are used for dewatering but these systems are not useful for pathogen reduction where thermal heat can play an important role. Thermal heat has a high operational cost. So, solar energy for thermal heat generation can be an economical and efficient solution. Dried sludge reduces the storage, handling, and transportation cost for sludge.

A case study on wastewater sludge drying was conducted by Belloulid et al., (2019) in Morocco. The greenhouse dryer was designed for sludge drying to reduce the water from sludge and results were compared with OSD. Fresh sludge samples with 80% initial MC were considered for the experiment. To reach the final MC of 20%, OSD took 45 h and 65 h whereas, the greenhouse dryer took 32 h and 57 h in summer and winter, respectively. Ameri et al., (2020) conducted a study in Algeria on the drying behavior of wastewater sludge in indirect and direct solar drying (ISD and DSD). Drying rate, moisture diffusion coefficient, and drying efficiency were analyzed and found that better performance on the ISD method than the DSD method. The average drying temperature was observed to be 43 and 41 °C in ISD and DSD, respectively. The diffusion coefficient was in the range of 1.91 to 9.12 × 10^{-8} m².s⁻¹. The activation energy for ISD and DSD dryers was 32.16 and 32.01 kJ mol⁻¹, respectively.

Mechanical dewatering is a method to separate sludge into a liquid and a solid part. This dewatered sludge need to be dried again to remove its MC. A study was performed by Salihoglu et al. (2007) to remove MC from the dewatered sludge with a plastic glazed solar tunnel greenhouse dryer. Solar drying after mechanical dewatering helped to reduce the sludge weight by 60% and the addition of 15% lime before drying helped to reduce the pathogen in the sludge. Solar dryer with rock bed as a TES material was used for sludge drying (Poblete and Painemal, 2020). Rock bed material helped to improve the drying process and reduced the drying time by providing the required temperature after sunshine hours. The energy efficiency of the dryer with the TES system was 21.68% higher than the dryer without TES.

Biogas plant waste slurry is a good natural fertilizer for agricultural land. Also, the dried slurry can be used for landfilling as it consists of a lot of natural nutrients. Drying of slurry is the major task in the biogas plant. If the plant was provided with necessary arrangements for solar drying, then the exhaust slurry can be properly treated with 15–20% of lime and can be dried. The drying time of the slurry was reduced to 65% compared to the OSD method (Kumar and Prakash, 2019). As the temperature requirement in wastewater drying is not more than 80 °C which is achievable with the help of greenhouse dryers, ISD, and mixed-mode dryers.

Once again proved that the ISD process is always better than the DSD process under the same working conditions (same solar radiation). Based on the evaluation of thermodynamic characteristics of drying of sludge, it is known that OSD creates an uneven drying, whereas, ISD produces better performance. This sector needs a further analysis on bacterial reaction before, during, and after the drying process and more studies are needed to concentrate on dewatering, stabilization, incineration, thickening, etc. so that the solar drying of sludge can be projected in a better way to industries and society.

4.9. Lignite coal industry

Lignite is the foremost fossil fuel used in thermal power plants for electricity generation which carries about 30–70% of MC. The MC creates further difficulty during the combustion process. It is necessary to remove the MC before feeding into the combustion chamber (Nikolopoulos et al., 2015). Liu et al., (2017) analyzed the performance of solar drying of lignite coal in a thermal power plant as shown in Fig. 4. A parabolic dish collector was used for the pre-drying of lignite coal and the dried lignite was fed into the boiler for combustion which helped to increase the boiler efficiency. It was observed that the overall combustion efficiency of lignite coal was influenced by the preheating of coal before feeding into the boiler. Also observed that both dryer efficiency and boiler efficiency influenced the solar-to-electric conversion efficiency of 34% was achieved for the proposed plant.

A solar dryer with flat plate air solar collectors was used for the predrying of lignite in a solar-lignite hybrid power generation plant (Xu et al., 2018). The dried lignite, using solar energy, has a high calorific value which helped to increase the plant efficiency. Thermal energy, exergy efficiency, and economic analysis were carried out to find the effect of dried lignite on system performance and lignite expenditure. The solar-dried lignite helped to increase the boiler efficiency by 1.2% which saved the lignite expenditure cost and increased the overall energy efficiency by 0.5%. Solar electrical efficiency was estimated to be 21.2%. A similar type of solar lignite hybrid power generation system was proposed by Xu et al. (2017). The effect of integration of solar energy for pre-drying of low-rank coal and low-temperature pyrolysis on system performance was investigated. It was observed that the solar drying of low-rank coal helped to achieve the required quality of coal for boiler feeding. The proposed system was integrated with a typical 600 MW supercritical power plant for further analysis. Overall energy and exergy efficiencies of the system were estimated to be 90% and 89.8%, respectively.

Han et al. (2020) proposed a solar lignite hybrid power generation system where solar-driven heat pumps were integrated with waste heat recovery. The system gave a two-stage drying for the lignite. Waste heat was absorbed by the solar-driven heat pump and utilized for pre-drying of lignite, which helped to increase the heating value of the fuel. Both thermodynamics and economic analysis were carried out for the 660 MW power plant. The authors claimed that the proposed method helped to generate the additional 50.3 MW power with 3.47% higher efficiency.

From the above studies, it is proved that solar energy is successfully used to dry lignite/coal. The coal/lignite is a huge quantity, sometimes in tonnes, compared to few kgs of food materials or other industrial applications. Drying of huge quantum (of lignite/coal) is a tough task and need expensive solar plants with huge collector area. Once it is made, the system works fine without further expenses. Also, the dryer gives enhanced performance parameters such as energy, exergy, solar-to-electric conversion, boiler efficiencies (Xu et al., 2017; Xu et al., 2018) when normal dryers were updated with solar dryers. Also, MC presents in lignite/coal gives more problems in coal generating unit and fuel handling equipment and hence this natural energy can be used to remove the MC.



Fig. 4. Schematic of lignite solar drying for preheating of lignite before firing (Liu et al., 2017).

From the above studies on various industries, it is observed that solar energy has huge potential in the area of renewable energy sources and it can be implemented effectively. Table 4 gives the temperature required and types of dryers used in various industries for drying operation.

5. Economic, environmental and social aspects

The investment cost of the solar dryer is more compared to the conventional method of drying (Bennamoun, 2011) but compared to traditional OSD drying, it offers several economic advantages such as reducing the fuel cost and emission from fuel. Also, a quality dried product increases its market value. Annualized costs, the life cycle of dryers, and payback period have been used to examine the economic feasibility of dryers but it depends on many factors such as drying material, geographical location, size, capacity, design factor, system efficiency, etc. (Kumar et al., 2016; Lingayat et al., 2020b). Also improved nutritional value, less space occupied by dryers, and higher yield also should be considered while doing the economic analysis. Emission from the fuel after combustion is responsible for global warming and pollution problem. Various policies and norms are implemented in different countries to reduce emissions.

The governments suggest running different industries with the help of solar energy and also give subsidies on solar-based systems (Pirasteh et al., 2014b). The power consumed by the fan in the forced greenhouse dryer is only 5% of the total energy required. If the fan is also run by solar energy or PV panels, then the total artificial energy required is zero (Mugi and Chandramohan, 2021). If conventional energy sources assisted with solar energy, then the dryers could save 20-40% of conventional fuel energy (Liu et al., 2015). Many researchers reported that the use of solar dryers could reduce CO2 emissions compared to other drving techniques (Ekechukwu and Norton, 1999; Hii et al., 2012). As per the report of (REN21, 2019), about 20% of the global energy is generated from renewable energy sources which included wind, solar, biomass, geothermal. Also, renewable policies create employment opportunities in the solar energy sector. The use of solar energy would significantly reduce the environmental pollution problem and improve public health.

There are some industries which already employed solar energy for different drying operations and are listed in Table 5.

6. Important findings

All the solar drying methods used in different industrial applications were explained in the previous sections. The important findings of each industrial application on solar drying are listed in Table 6. There were many agricultural materials such as turmeric, red chili, bitter gourd, garlic, apricot, apple, orange, black turmeric, cassava, ghost chili pepper, cherry tomato, pineapple, rice, etc. were dried in a solar dryer as discussed in this article (section 4.1). Solar dried turmeric gave good results in the form of its color, phenolic content, and flavonoid content. The heat losses such as convection and radiation must be reduced to increase the system performance. Paraffin wax was used as a TES material in most of the studies in food drying applications.

The marine foods such as shrimps, salted silver jewfish, queenfish, and Tilapia fish were dried in solar dryers as discussed in section 4.2. Fish flesh is a very sensitive tissue so that it needs to dry continuously to avoid wastage. Therefore, along with solar energy, other auxiliary sources of energy such as biomass, industrial waste heat, diesel burner, etc. must be used to dry the tissues round-a-clock so that a better final product could be achieved. The studies available for solar drying of the tea industry are chamomile tea leaves (Amer et al., 2018) and Mexican tea leaves (Kane et al., 2008) and they showed that the solar-dried tea leaves gave good results in terms of color and flavor. Also, the hybrid solar dryers reduced the drying time up to 20 h (section 4.3).

Paint curing and polymer coating are the applications of solar drying in the automobile industry. The paint curing needs an air temperature range of 80 to 150 °C which is possible using solar energy (Giampieri et al., 2020). The polymer coating is used on the steel surfaces which is used to avoid corrosion. This polymer coating needs to be effectively dried and the solar drying concept can be preferred for this application (section 4.4). 80% low-quality rubber is produced worldwide because of improper drying and therefore, these sheets are sold for low market price (Breymayer et al., 1993). Smoke drying is the preferred drying method that needs to be used for rubber sheet drying. The temperature range maintained in the solar dryer is 45 to 60 °C. Solar dryer reduced the drying time of rubber sheets up to 3 days compared to the OSD method (section 4.5).

The developed setup for paper drying (Madhavan and Ramachandran, 2015) gives a pay-back period of 95 days only and it's an alarming fact that all possible options of solar paper drying trials need to be encouraged (section 4.6). The sugarcane industry is another sector using solar energy to dry the crushed leaves (bagasse) and waste stem. They are used for biomass and landfilling purposes as they have good nutrients and importantly used for bio-ethanol production as mentioned in section 4.7.

Solar energy is used in sewage and wastewater treatment as dewatering cannot eliminate the bacteria in the sludge. As the contents are huge, it is better to use solar energy to dry the sludge. The dried solid

A. Lingayat et al.

Table 4

The temperature required and various types of dryers used in various industries during the drying processes.

Table 5

• •

Industries using solar energy for drying applications (Solar Termal Heat for Industrial Processes (SHIP), 2019).

Sector	Process	Temperature Range (°C)	Type of solar dryers used	1			
Agricultural and food industries	Drying	40–200	 Almost all types of natural and force convection solar dryers with and without TES 				
Marine industries	Drying	40–80 °C	 Solar water heating system with air–water heat exchanger. Indirect type hybrid FCSD dryer Fudholi and Sopian, (2019) 				
			 Solar tunnel dryer (Bala and Mondol, 2001). Hybrid solar dryer (Hamdani et al. 2018). 				
Tea industries	Drying of tea leaf	40–70 °C	Vacuum-assisted solar dryers (Pou and Tripathy 2020)				
			 Indirect type FCSD with PV-T collector (Ozturk and Dincer, 2019) Indirect type natural 				
			 narcet type haldrafter convection solar dryer (Amer et al., 2018) Roof-integrated solar dryer (Palaniappan and Subramanian, 1998) 				
			• FCSD dryer (Kane et al., 2008).				
Automobile industry	Surface treatment Drying of finished	20–120 60–150	 Solar water heating system with air-water heat exchanger. (Giampieri et al. 2020) 	I			
Rubber industry	components Drying of rubber sheets	45–150	Orsato and Wells, 2007).ISD with SAC (Pratoto et al., 1998).	(
-	Preheating	50–70	 Mixed-mode solar dryer (Dejchanchaiwong et al., 2016) Greenhouse solar dryer (Janjai et al. 2015; Jitjack 	I			
Pulp, paper and allied industries	Drying	45–80	et al., 2016) • Indirect type natural convection solar dryer (Madhavan and Ramachandran, 2015) • Solar water heating				
			system with air–water heat exchanger (Hjort and Thomas, 2014).				
Sugarcane industry	Drying of bagasse	40–65	 FCSD (Phadkari et al. 2017) Solar tunnel dryer (Subahana and Natarajan, 2016) 	dr wl va be			
Sewage and industrial waste	_	40–65	 Greenhouse dryer (Belloulid et al., 2019: Salihoglu et al. 2007) Mixed-mode dryers. ISD and DSD (Ameri et al., 	be is re: ty			
Lignite coal industry	~70	Pre-drying of lignite coal	2020) Parabolic dish collector (Liu et al., 2017; Xu et al., 2018)	11a co pa			

waste can be used for the landfill. Also, the storage, transportation, and handling costs of dried sludges were low (section 4.8) compared to raw wastes. MC in lignite/coal reduces the combustion characteristics and also create problem in lignite/coal units. The solar-dried lignite/coal produced high fuel value and the boiler efficiency was improved as explained in section 4.9.

muusury	name	Process	collector	(°C)
Agriculture and Food	Coopeldos, Costa rica	Coffee drying	Solar air heater	40–45
	Duren Coffe, Panama Duren	Coffee drying	Solar air collector	40–45
	Keyaqa Orchards, USA	Walnuts drying	Solar air collector	~43
	Gengli Fruit Drying, China	Fruit drying	Solar air collector	50–70
	Malabar Tea Drying, Indonesia	Tea drying	Solar air collector	~35
	Carriers and Sons, United state	Food drying	Solar air collector	~43
	AMR dal Mill, India	Drying of pulse	Solar air collector	65–75
	Aroma Plant Romania, Romania Aroma	Drying of medicinal plants	Solar air collector	
	Sonoma Country Herb Exchange, USA	Drying of herbs	Solar air collector	
	Grammer Solar Argentina, Argentina	Drying of tobacco	Solar air collector	
	India Tobacco Division, India	Conditioning of tobacco	Solar air collector	~95
Rubber and Plastic Products	Inter Rubber Latex Co. Ltd., Thailand	Drying of natural rubber	Solar air collector	
Concrete	Leitt Beton Gmbh, Austria	Drying of pre- fabricated concrete components	Solar air collector	
Furniture	Carpenting Hamminger, Austria	Wood Drying	Flat plate collector	25–115
Mining	Korner Kvk, Austria	Preheating, drying of raw material	Flat plate collector	50-80

7. Recommendations proposed for better performance

After predicting the potential of solar energy for different industrial rying applications, it is important to propose the solar dryer system, hich should be simple in design, cost-effective, and suitable for drying arious food and other products. Different techniques and methods have een suggested and implemented for a reduction in drying time and etter product quality. The assistance of a solar water heater with dryers one of the possible solutions to make the hybrid drying systems using newable energy. Various types of solar water heaters of concentrating pe and non-concentrating types are available in the market. They are; at plate collectors, evacuated tube collectors, compound parabolic llectors, parabolic trough collectors, cylindrical trough collectors, parabolic dish reflectors, heliostat field reflectors, linear Fresnel reflectors, etc. These collectors can provide a temperature range of 30 to 500 °C (Mekhilef et al., 2020; Pranesh et al., 2019). The assistance of solar drying systems with a water storage system and water-air heat exchanger can make the dryer more efficient for a wide range of applications in the area of solar drying Fig. 5 shows a generalized technic for thermal energy conservation that can be implemented for various drying applications. Solar collectors are used to absorbing solar radiation. Fluid flowing through the collectors absorbs the heat from the collector and is stored in the tank. The heat energy retrieved from the

Solar Energy xxx (xxxx) xxx

Table 6 Important findings from the literature. s. Authors (year) Research outcomes and findings No Agricultural and food industries 1 Lakshmi et al. (2018) 2 Ndukwu et al. (2017) 3 Zachariah et al. (2020)

A. Lingayat et al.

4

5

6

2

Drying air temperature is the highest (65 °C) among all other research works. The collector efficiency is low (12%). Dryer with TES material of Na₂SO₄·10H₂O consumed energy of 3.34 kWh per kg removal of MC from the products. NCSD integrated with TES effectively minimized various heat losses which lead to efficient energy utilization. A mixed-mode solar dryer integrated with PCM produced an increase in η_d and reduction in drving time. It took 10 h time to completely dry the product. Banjasadi et al. (2017) At the end of the drying process, 25% of MC is available inside the product (apricot slices) Add 2 more important results. Atalay et al. (2017) Effective utilization of waste heat (about 50 to 60%) is possible with this system The drying process took 6 h to completely dry the apple slices. Atalay (2019) Improved drying efficiency (34%) and SMER of 0.505 kg of water/kWh New parameters such as waste exergy ratio, sustainability index, and improvement potential were introduced. Energy and exergy efficiencies decreased up to 68.37%. Marine industries Murali et al. (2020) Drying shrimps is a very fast process compared to all other drying products. It took only 6 h time to dry 50 kg of shrimps. Drying and collector efficiencies were also improved up to 37.09% and 42.37%, respectively. A new drying method (solar tunnel dryer) was Bala and Mondol (2001) introduced to dry the salted silver Jew fish. The setup saved the product from rain, dirt, dust, and insects and also maintained the

The drying efficiency of the system is improved up to 29% using a hybrid drying system (solar

+ auxiliary diesel burner) to dry salted silver

reduced to 2.92 kWh / kg of moisture removal.

The cost incurred to fabricate a hybrid solar

year, selling price \$ 3.3 / kg, obtained IRR =

Solar dryer with reflector, thermal storage

The roof-integrated solar drver can be used

along with conventional energy sources to

The paint drying in automobile surface is

performed with a temperature limit of 50 to

compared to solvent-based paints (2 to 5 h).

A mixed-mode solar dryer produced better

results for rubber sheet drying than the ISD

better quality of natural rubber.

Water-based paints took 3 to 8 h of drving time

Smoke-assisted solar drying can be used for the

reduce the annual fuel consumption by about

system and two-chamber dryer can be used to

increase the drying rate and capacity of dryer.

18.61%, NPV = \$ 21.091 and break-even point

The specific energy consumption (SEC) is

dryer to dry 100 kg of fish is \$ 1870. Dry fish production capacity was 12,000 kg per

3 Fudholi and Sopian (2019) quality

Jew fish.

= 2.6 years.

25%.

80 °C.

dryer.

4 Hamdani et al. (2018)

Tea industries

Amer et al. (2018) 1

2 Palaniappan and Subramanian (1998)

Automobile industry

Giampieri et al. (2020) 1

Rubber industry

Breymayer et al. (1993) 1

2 Dejchanchaiwong et al. (2016)

3 Janjai et al. (2015). Table 6 (continued)

S. No	Authors (year)	Research outcomes and findings
		A greenhouse dryer can be used for the mass production of dried rubber sheets.
Pulp,	paper and allied industries	
1	Madhavan and Ramachandran (2015)	Handmade paper is dried in a solar dryer. It helped to reduce the drying time and gave better quality paper compared to OSD. Solar dryers were functional during day time
2	Hjort and Thomas (2014)	even there is no sunlight and more humidity. A solar-heated hot water flat tank could be a better and efficient option for paper drying. Experiments were performed for different operating conditions and paper types. It can be made functional in larger-scale dryers.
Sugar	cane industry	
1	Subahana and Natarajan (2016)	A tunnel dryer with TES material of soil and preheated air gave a better moisture removal rate than a simple tunnel dryer.
2	Vijayaraj and Saravanan (2008)	The solution was used to approximate and quantify the different heat and mass transfer phenomena in bagasse for different conditions of drying.
Sewag	ge and industrial waste	
1	Ameri et al. (2020)	ISD process is always better than DSD process under the same working conditions. Thermodynamic characteristics of the drying of sludge process were introduced to find out the nature and disorderness of drying process.
2	Belloulid et al. (2019)	The developed solar dryer for sludge and waste is fit for arid regions. The drying time saved in the solar dryer is 13 h
Linnia	a a a l in du atum	in summer and 8 n in winter compared to OSD
	Lin et al. (2017)	Draing of lignite (goal with color operation
2	Xu et al. (2018)	increased the fuel value and the overall combustion efficiency of lignite coal. The solar-to-electric conversion efficiency was more than 34%. Solar-lignite hybrid power generation system
-		helped to increase the boiler efficiency by 1.2% and saved lignite expenditure cost. Overall energy efficiency increased by 0.5%.

hot water (from the tank) transferred to the drying chamber and the materials can be dried.

This approach can be explored in all drying applications in industries. Integration of efficient utilization of solar energy with other renewable energy systems such as photovoltaic-thermal, dryer with TES system, solar and biofuel can improve the efficiency of the dryers. To make the solar drying system 100% renewable, it can be assisted with a ground source heat pump system. (Esen, 2006; Esen et al., 2007a) This needs detailed research in depth which can provide a positive impact on drying efficiency. Already a lot of mathematical (analytical and numerical) studies are available but most of the experimental studies are limited up to laboratory scale. So, it's another research area where solar dryers can be explored on large scale for enhancement of production capacities.

A cost-effective approach by considering the capital and operational costs along with solar-based technologies is necessary for the effective implementation of dryers. The PCM is found to be a better option in the TES system of solar dryers. The efficiency of solar dryers can be improved by adding nano-materials in PCM which is used to increase the thermal conductivity of PCM. Even though the solar energy approach looks simple, a deep understanding of complex drying operations is necessary for large-scale applications for the success of the developed drying systems. The automation of the complete drying process can improve the quality of material and drying efficiency. By considering all these advantages of solar-based drying systems, the government and private sectors should encourage the researchers.



Fig. 5. Solar drying systems assisted with solar water heater and water-air heat exchanger.

8. Conclusions

Solar drying concept is an emerging technology for drying industries such as the food, automobile, paper, and allied products, rubber, sugarcane, sewage and industrial waste. Available literature reviews on solar dryers contributed more to the agricultural sector. This work reviewed the contribution of solar dryers of all industries. The presented review focused on industries where drying is involved during their production processes. Industrial drying process systems were analyzed concerning the type of material, existing solar drying technologies, required temperatures for drying, etc.

The required drying period of the product was different for different industries and their materials. The essential drying factors for all the industries were moisture content (MC) of the product, temperature, velocity, humidity of drying air, and geographical location. The solar dryers assisted with thermal energy storage (TES) systems helped to decrease the total time required for drying. Also, the addition of the TES system increased the system performance, energy, exergy, solar-toelectric conversion efficiencies. The use of hybrid solar dryers (such as dryers with LPG water heaters, diesel burners, and biomass-fueled air heaters) helped to satisfy the electrical power requirement so that the dryer could work round-a-clock. Forced convection solar dryers (FCSD) gave better performance compared to NCSD setups, but they needed little artificial energy for running the fan or blower. FCSD setups could run with zero electrical power using solar photovoltaic panels to run fans or blowers.

Solar-based drying technology is a promising area of research. The commercialization of solar dryers is increasing day by day for different drying applications in industries. More researches on different applications further help industries to make their setups so that society can enjoy its outcomes. This article helps scientists and researchers for finding the opportunity in solar-based drying systems for different industries and making feasible drying operations.

Declaration of Competing Interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgments

The authors wish to acknowledge the support received by way of proofreading from Dr. M. R. Vishwanathan, Assistant Professor of English, Department of Humanities and Social Science, NIT Warangal, India.

References

- Amer, B.M.A., Gottschalk, K., Hossain, M.A., 2018. Integrated hybrid solar drying system and its drying kinetics of chamomile. Renew. Energy 121, 539–547. https://doi.org/ 10.1016/j.renene.2018.01.055.
- Ameri, B., Hanini, S., Boumahdi, M., 2020. Influence of drying methods on the thermodynamic parameters, effective moisture diffusion and drying rate of wastewater sewage sludge. Renew. Energy 147, 1107–1119. https://doi.org/ 10.1016/j.renene.2019.09.072.
- Arunsandeep, G., Lingayat, A., Chandramohan, V.P., Raju, V.R.K., Reddy, K.S., 2018. A numerical model for drying of spherical object in an indirect type solar dryer and estimating the drying time at different moisture level and air temperature. Int. J. Green Energy 15, 189–200. https://doi.org/10.1080/15435075.2018.1433181.
- Atalay, H., 2019. Performance analysis of a solar dryer integrated with the packed bed thermal energy storage (TES) system. Energy 172, 1037–1052. https://doi.org/ 10.1016/j.energy.2019.02.023.
- Atalay, H., Turhan Çoban, M., Kıncay, O., 2017. Modeling of the drying process of apple slices: Application with a solar dryer and the thermal energy storage system. Energy 134, 382–391. https://doi.org/10.1016/j.energy.2017.06.030.
- Ayensu, A., 1997. Dehydration of food crops using a solar dryer with convective heat flow. Sol. Energy 59, 121–126. https://doi.org/10.1007/978-1-4614-5806-7.
- Ayyappan, S., Mayilsamy, K., Sreenarayanan, V.V., 2016. Performance improvement studies in a solar greenhouse drier using sensible heat storage materials. Heat Mass Transf. 52, 459–467. https://doi.org/10.1007/s00231-015-1568-5.
- Bala, B.K., Mondol, M.R.A., 2001. Experimental investigation on solar drying of fish using solar tunnel dryer. Dry. Technol. An Int. J. 19, 427–436. https://doi.org/ 10.1081/DRT-100102915.
- Baniasadi, E., Ranjbar, S., Boostanipour, O., 2017. Experimental investigation of the performance of a mixed-mode solar dryer with thermal energy storage. Renew. Energy 112, 143–150. https://doi.org/10.1016/j.renene.2017.05.043.
- Belloulid, M.O., Hamdi, H., Mandi, L., Ouazzani, N., 2019. Solar drying of wastewater sludge: a case study in Marrakesh. Morocco. Environ. Technol. (United Kingdom) 40, 1316–1322. https://doi.org/10.1080/09593330.2017.1421713.
- Benli, H., Durmuş, A., 2009. Performance analysis of a latent heat storage system with phase change material for new designed solar collectors in greenhouse heating. Sol. Energy 83, 2109–2119. https://doi.org/10.1016/j.solener.2009.07.005.
- Bennamoun, L., 2011. Reviewing the experience of solar drying in Algeria with presentation of the different design aspects of solar dryers. Renew. Sustain. Energy Rev. 15, 3371–3379. https://doi.org/10.1016/j.rser.2011.04.027.
- Bennamoun, L., Belhamri, A., 2003. Design and simulation of a solar dryer for agriculture products. J. Food Eng. 59, 259–266. https://doi.org/10.1016/S0260-8774(02) 00466-1.
- Breymayer, M., Pass, T., Mühlbauer, W., Amir, E.J., Mulato, S., 1993. Solar-assisted smokehouse for the drying of natural rubber on small-scale Indonesian farms. Renew. Energy 3, 831–839. https://doi.org/10.1016/0960-1481(93)90039-J.
- Celma, A.R., Cuadros, F., 2009. Energy and exergy analyses of OMW solar drying process. Renew. Energy 34, 660–666. https://doi.org/10.1016/j. renene 2008.05.019
- Chandramohan, V.P., 2016a. Experimental analysis and simultaneous heat and moisture transfer with coupled CFD model for convective drying of moist object. Int. J. Comput. Methods Eng. Sci. Mech. 17, 59–71. https://doi.org/10.1080/ 15502287.2016.1147506.
- Chandramohan, V.P., 2016b. Numerical prediction and analysis of surface transfer coefficients on moist object during heat and mass transfer application. Heat Transf. Eng. 37, 53–63. https://doi.org/10.1080/01457632.2015.1042341.
- Chandramohan, V.P., Talukdar, P., 2016. Estimation of equilibrium moisture content and drying time of potato during convective drying. Fluid Mech. Fluid Power - Lect. Notes. Mech. Eng. 205–213 https://doi.org/10.1007/978-81-322-2743-4_21.

A. Lingayat et al.

Chandramohan, V.P., Talukdar, P., 2013. Design of an experimental set up for convective drying: experimental studies at different drying temperature. Heat Mass Transf. 49, 31–40. https://doi.org/10.1007/s00231-012-1060-4.

Chang, K., 2015. World tea production and trade. current and future development. Food and Agriculture Organization of the United Nations.

Dejchanchaiwong, R., Arkasuwan, A., Kumar, A., Tekasakul, P., 2016. Mathematical modeling and performance investigation of mixed-mode and indirect solar dryers for natural rubber sheet drying. Energy Sustain. Dev. 34, 44–53. https://doi.org/ 10.1016/j.esd.2016.07.003.

Dhalsamant, K., Tripathy, P.P., Shrivastava, S.L., 2018. Heat transfer analysis during mixed-mode solar drying of potato cylinders incorporating shrinkage: Numerical simulation and experimental validation. Food Bioprod. Process. 109, 107–121. https://doi.org/10.1016/j.fbp.2018.03.005.

Ekechukwu, O.V., Norton, B., 1999. Review of solar - energy drying systems II : an overview of solar drying technology. Energy Convers. Manag. 40, 615–655. https:// doi.org/10.1016/S0196-8904(98)00092-2.

El-Sebaii, A.A., Shalaby, S.M., 2013. Experimental investigation of an indirect-mode forced convection solar dryer for drying thymus and mint. Energy Convers. Manag. 74, 109–116. https://doi.org/10.1016/j.enconman.2013.05.006.

El-sebaii, A.A., Shalaby, S.M.M., 2012. Solar drying of agricultural products : A review. Renew. Sustain. Energy Rev. 16, 37–43. https://doi.org/10.1016/j. rser.2011.07.134.

El Khadraoui, A., Bouadila, S., Kooli, S., Farhat, A., Guizani, A., 2017. Thermal behavior of indirect solar dryer: Nocturnal usage of solar air collector with PCM. J. Clean. Prod. 148, 37–48. https://doi.org/10.1016/j.jclepro.2017.01.149.

Elhage, H., Herez, A., Ramadan, M., Bazzi, H., Khaled, M., 2018. An investigation on solar drying : A review with economic and environmental assessment. Energy. https://doi.org/10.1016/j.energy.2018.05.197.

ELkhadraoui, A., Kooli, S., Hamdi, I., Farhat, A., 2015. Experimental investigation and economic evaluation of a new mixed-mode solar greenhouse dryer for drying of red pepper and grape. Renew. Energy 77, 1–8. https://doi.org/10.1016/j. renene.2014.11.090.

Embong, R., Shafiq, N., Kusbiantoro, A., Nuruddin, M.F., 2016. Effectiveness of lowconcentration acid and solar drying as pre-treatment features for producing pozzolanic sugarcane bagasse ash. J. Clean. Prod. 112, 953–962. https://doi.org/ 10.1016/j.jclepro.2015.09.066.

Esen, H., 2006. Technoeconomic appraisal of a ground source heat pump system for a heating season in eastern Turkey. Energy Convers. Manag. 47, 1281–1297. https:// doi.org/10.1016/j.enconman.2005.06.024.

Esen, H., Inalli, M., Esen, M., 2007a. A techno-economic comparison of ground-coupled and air-coupled heat pump system for space cooling. Build. Environmen 42, 1955–1965. https://doi.org/10.1016/j.buildenv.2006.04.007.

Esen, H., Inalli, M., Esen, M., Pihtili, K., 2007b. Energy and exergy analysis of a groundcoupled heat pump system with two horizontal ground heat exchangers. Build. Environmen 42, 3606–3615. https://doi.org/10.1016/j.buildenv.2006.10.014.

Esen, H., Ozgen, F., Esen, M., Sengur, A., 2009a. Expert Systems with Applications Artificial neural network and wavelet neural network approaches for modelling of a solar air heater. Expert Syst. Appl. 36, 11240–11248. https://doi.org/10.1016/j. eswa.2009.02.073.

Esen, H., Ozgen, F., Esen, M., Sengur, A., 2009b. Expert Systems with Applications Modelling of a new solar air heater through least-squares support vector machines. Expert Syst. with Appl. J. 36, 10673–10682. https://doi.org/10.1016/j. eswa.2009.02.045.

Esen, M., 2000. Thermal Performance of a Solar-Aided Latent Heat Store used for Space Heating by Heat Pump. Sol. Energy 69, 15–25.

Esen, M., Ayhan, T., 1996. Development of a Model Compatible with Solar Assisted Cylindrical Energy Storage Tank and Variation of Stored Energy with Time for Different Phase Change Materials. Energy Convers. Mgm 37, 1775–1785. https:// doi.org/10.1016/0196-8904(96)00035-0.

Esen, M., Durmus, A., 1998. Geometric Design of Solar-Aided Latent Heat Store Depending on Various Parameters and Phase Change Materials. Sol. Energy 62, 19–28. https://doi.org/10.1016/S0038-092X(97)00104-7.

Esen, M., Yuksel, T., 2013. Experimental evaluation of using various renewable energy sources for heating a greenhouse. Energy Build. 65, 340–351. https://doi.org/ 10.1016/j.enbuild.2013.06.018.

Fudholi, A., Sopian, K., 2019. A review of solar air flat plate collector for drying application. Renew. Sustain. Energy Rev. 102, 333–345. https://doi.org/10.1016/j. rser.2018.12.032.

Fudholi, A., Sopian, K., Bakhtyar, B., Gabbasa, M., Yusof, M., Ha, M., 2015. Review of solar drying systems with air based solar collectors in Malaysia. Renew. Sustain. Energy Rev. 51, 1191–1204. https://doi.org/10.1016/j.rser.2015.07.026.

Gemechu, E.D., Butnar, I., Llop, M., Castells, F., 2012. Environmental tax on products and services based on their carbon footprint: A case study of the pulp and paper sector. Energy Policy 50, 336–344. https://doi.org/10.1016/j.enpol.2012.07.028.

Giampieri, A., Ling-Chin, J., Ma, Z., Smallbone, A., Roskilly, A.P., 2020. A review of the current automotive manufacturing practice from an energy perspective. Appl. Energy 261. https://doi.org/10.1016/j.apenergy.2019.114074.

Hamdani, Rizal, T.A., Muhammad, Z., 2018. Fabrication and testing of hybrid solarbiomass dryer for drying fish. Case Stud. Therm. Eng. 12, 489–496. https://doi.org/ 10.1016/j.csite.2018.06.008.

Han, Y., Sun, Y., Wu, J., 2020. An efficient solar / lignite hybrid power generation system based on solar- driven waste heat recovery and energy cascade utilization in lignite pre-drying. Energy Convers. Manag. 205, 112406 https://doi.org/10.1016/j. enconman.2019.112406.

Hii, C.L., Jangam, S.V., Ong, S.P., Mujumdar, A.S., 2012. Solar Drying: Fundamentals. Applications and Innovations. Publ. Singapore 1–176. Hjort, M., Thomas, D.E., 2014. Performance of a solar paper drier for small-scale paper sheet production. J. Mech. Eng. 44, 6–9. https://doi.org/10.3329/jme.v44i1.19491.

Solar Energy xxx (xxxx) xxx

Hossain, M.A., Amer, B.M.A.A., Gottschalk, K., 2008. Hybrid solar dryer for quality dried tomato. Dry. Technol. 26, 1591–1601. https://doi.org/10.1080/ 07373930802467466.

International Energy Outlook (IEO) 2019, U. S. Energy Information Administration, 2019. , U.S. Department of Energy. https://doi.org/10.5860/choice.44-3624.

- Janjai, S., Piwsaoad, J., Nilnont, W., Pankaew, P., 2015. Experimental performance and neural network modeling of a large-scale greenhouse solar dryer for drying natural rubber sheets. J. Control Sci. Eng. 3, 48–53. https://doi.org/10.17265/2328-2231/ 2015.01.006.
- Jitjack, K., Thepa, S., Sudaprasert, K., Namprakai, P., 2016. Improvement of a rubber drying greenhouse with a parabolic cover and enhanced panels. Energy Build. 124, 178–193. https://doi.org/10.1016/j.enbuild.2016.04.030.
- K. Sopian, M.Y. Othman, B. Yatim, M.H.R., 2000. Experimental studies on a solar assisted drying system for herbal tea. World Renew. Energy Congr. VI 2000 Bright. UK 1139–1142.
- Kalogirou, S.A., Karellas, S., Braimakis, K., Stanciu, C., 2016. Exergy analysis of solar thermal collectors and processes 56, 106–137. https://doi.org/10.1016/j. pecs.2016.05.002.
- Kane, C.S.E., Jamali, A., Kouhila, M., Mimet, A., Ahachad, M., 2008. Single-layer drying behavior of Mexican tea leaves (Chenopodium ambrosioides) in a convective solar dryer and mathematical modeling. Chem. Eng. Commun. 195, 787–802. https://doi. org/10.1080/00986440701691095.

Knaack, J., Klein, P., Surridge, K., Wolf, M., Horta, P., 2018. Solar heat for industry. German Solar Association. doi 10 (4135/9781412953979), n572.

Kumar, M., Sansaniwal, S.K., Khatak, P., 2016. Progress in solar dryers for drying various commodities. Renew. Sustain. Energy Rev. 55, 346–360. https://doi.org/10.1016/j. rser.2015.10.158.

Kumar, S., Prakash, O., 2019. Biogas plant slurry dewatering and drying using hybrid system: A review, Renewable Energy and its Innovative Technologies. Springer, Singapore. https://doi.org/10.1007/978-981-13-2116-0.

- Lakshmi, D.V.N., Muthukumar, P., Layek, A., Nayak, P.K., 2018. Drying kinetics and quality analysis of black turmeric (Curcuma caesia) drying in a mixed mode forced convection solar dryer integrated with thermal energy storage. Renew. Energy 120, 23–34. https://doi.org/10.1016/j.renene.2017.12.053.
- Lamidi, R.O., Jiang, L., Pathare, P.B., Wang, Y.D., Roskilly, A.P., 2019. Recent advances in sustainable drying of agricultural produce: A review. Appl. Energy 233–234, 367–385. https://doi.org/10.1016/J.APENERGY.2018.10.044.
- Lingayat, Abhay, Chandramohan, V.P., R.K., R. V., Suresh, S., 2020a. Drying kinetics of tomato (Solanum lycopersicum) and Brinjal (Solanum melongena) using an indirect type solar dryer and performance parameters of dryer. Heat Mass Transf. https://doi. org/doi.org/10.1007/s00231-020-02999-3.

Lingayat, A., Chandramohan, V.P., Raju, V.R.K., 2018. Numerical analysis on solar air collector provided with artificial square shaped roughness for indirect type solar dryer. J. Clean. Prod. 190, 353–367. https://doi.org/10.1016/j.jclepro.2018.04.130.

Lingayat, A., Chandramohan, V.P., Raju, V.R.K., 2017. Design, development and performance of indirect type solar dryer for banana drying. Energy Procedia 109, 409–416. https://doi.org/10.1016/j.egypro.2017.03.041.

Lingayat, Abhay, Chandramohan, V.P., Raju, V.R.K., Kumar, A., 2020b. Development of indirect type solar dryer and experiemnts for estimation of drying parameters of apple and watermelon. Therm. Sci. Eng. Prog. 16, 100477.

Lingayat, A.B., Chandramohan, V.P., Raju, V.R.K., Meda, V., 2020c. A review on indirect type solar dryers for agricultural crops – Dryer setup, its performance, energy storage and important highlights. Appl. Energy 258, 114005. https://doi.org/10.1016/j. apenergy.2019.114005.

Liu, M., Wang, C., Han, X., Li, G., Chong, D., Yan, J., 2017. Lignite drying with solar energy: Thermodynamic analysis and case study. Dry. Technol. 35, 1117–1129. https://doi.org/10.1080/07373937.2016.1233113.

Liu, M., Wang, S., Li, K., 2015. Study of the Solar Energy Drying Device and Its Application in Traditional Chinese Medicine in Drying. Int. J. Clin. Med. 06, 271–280. https://doi.org/10.4236/ijcm.2015.64034.

Madhavan, S., Ramachandran, P.N., 2015. Design, fabrication and testing of a solar paper dryer. Int. Res. J. Eng. Technol. 2, 1911–1914.

Malekjani, N., Jafari, S.M., 2018. Simulation of food drying processes by Computational Fluid Dynamics (CFD); recent advances and approaches. Trends Food Sci. Technol. 78, 206–223. https://doi.org/10.1016/j.tifs.2018.06.006.

Mathioudakis, V.L., Kapagiannidis, A.G., Athanasoulia, E., Diamantis, V.I., Melidis, P., Aivasidis, A., 2009. Extended Dewatering of Sewage Sludge in Solar Drying Plants. Desalination 248, 733–739. https://doi.org/10.1016/j.desal.2009.01.011.

Desalination 248, 733–739. https://doi.org/10.1016/j.desal.2009.01.011. Mekhilef, S., Saidur, R., Safari, A., 2020. A review on solar energy use in industries. Renew. Sustain. Energy Rev. 15, 1777–1790. https://doi.org/10.1016/j. rser.2010.12.018.

Mezrhab, A., Elfarh, L., Naji, H., Lemonnier, D., 2010. Computation of surface radiation and natural convection in a heated horticultural greenhouse. Appl. Energy 87, 894–900. https://doi.org/10.1016/j.apenergy.2009.05.017.

Mishra, S., Verma, S., Chowdhury, S., Dwivedi, G., 2020a. Analysis of recent developments in greenhouse dryer on various parameters- a review. Mater. Today Proc. https://doi.org/10.1016/j.matpr.2020.07.429.

Mohana, Y., Mohanapriya, R., Anukiruthika, T., Yoha, K.S., Moses, J.A., Anandharamakrishnan, C., 2020. Solar dryers for food applications: Concepts, designs, and recent advances. Sol. Energy 208, 321–344. https://doi.org/10.1016/j. solener.2020.07.098.

Mohanraj, M., 2014. Performance of a solar-ambient hybrid source heat pump drier for copra drying under hot-humid weather conditions. Energy Sustain. Dev. 23, 165–169. https://doi.org/10.1016/j.esd.2014.09.001.

A. Lingayat et al.

 Mohanraj, M., Chandrasekar, P., 2009. Performance of a forced convection solar drier integrated with gravel as heat storage material. J. Eng. Sci. Technol. 4, 305–314.
 Monthly Energy Review January 2020, U. S. Energy Information Administration, 2020.

- Mudiny Energy neversion standary 2020, 0. S. Energy information radiantication, 2020. Mugi, V.R., Chandramohan, V.P., 2021. Energy and exergy analysis of forced and natural convection indirect solar dryers: Estimation of exergy inflow, outflow, losses, exergy efficiencies and sustainability indicators from drying experiments. J. Clean. Prod. 282 https://doi.org/10.1016/j.jclepro.2020.124421.
- Muhlbauer, W., 1986. Present status of solar crop drying. Energy Agric. 5, 121–137. https://doi.org/10.1016/0167-5826(86)90013-6.
- Mujumdar, A.S., 2006. Hand book of Industrial drying, 3rd Ed. CRC Press, pp. 1–1312. Murali, S., Amulya, P.R., Alfiya, P.V., Delfiya, D.S.A., Samuel, M.P., 2020. Design and performance evaluation of solar - LPG hybrid dryer for drying of shrimps. Renew.
- Energy 147, 2417–2428. https://doi.org/10.1016/j.renene.2019.10.002. Mustayen, A.G.M.B., Mekhilef, S., Saidur, R., 2014. Performance study of different solar dryers : A review. Renew. Sustain. Energy Rev. 34, 463–470. https://doi.org/ 10.1016/j.rser.2014.03.020.
- Nabnean, S., Janjai, S., Thepa, S., Sudaprasert, K., Songprakorp, R., Bala, B.K., 2016. Experimental performance of a new design of solar dryer for drying osmotically dehydrated cherry tomatoes. Renew. Energy 94, 147–156. https://doi.org/10.1016/ i.renene.2016.03.013.
- Ndukwu, M.C., Bennamoun, L., Abam, F.I., Eke, A.B., Ukoha, D., 2017. Energy and exergy analysis of a solar dryer integrated with sodium sulfate decahydrate and sodium chloride as thermal storage medium. Renew. Energy 113, 1182–1192. https://doi.org/10.1016/j.renene.2017.06.097.
- Nikolopoulos, N., Violidakis, I., Karampinis, E., Agraniotis, M., Bergins, C., Grammelis, P., Kakaras, E., 2015. Report on comparison among current industrial scale lignite drying technologies (A critical review of current technologies). Fuel 155, 86–114. https://doi.org/10.1016/j.fuel.2015.03.065.
- Orsato, R.J., Wells, P., 2007. The Automobile Industry and Sustainability. J. Clean. Prod. 15, 989–993. https://doi.org/10.1016/j.jclepro.2006.05.035.
- Ozgen, F., Esen, M., Esen, H., 2009. Experimental investigation of thermal performance of a double-flow solar air heater having aluminium cans. Renew. Energy 34, 2391–2398. https://doi.org/10.1016/j.renene.2009.03.029.
- Ozturk, M., Dincer, I., 2019. Exergoeconomic analysis of a solar assisted tea drying system. Dry. Technol. 1–8. https://doi.org/10.1080/07373937.2019.1669044.
- Palaniappan, C., Subramanian, S.V., 1998. Economics of solar air pre-heating in South Indian tea factories: a case study. Sol. Energy 63, 31–37.
- Panwar, N.L., Kaushik, S.C., Kothari, S., 2012. A review on energy and exergy analysis of solar dying systems. Renew. Sustain. Energy Rev. https://doi.org/10.1016/j. rser.2012.02.053.
- Phadkari, S., Patil, S., Deokar, S.U., 2017. Design and Modeling of Solar Bagasse Dryer, in: International Conference on Ideas, Impact and Innovation in Mechanical Engineering (ICIIIME 2017). pp. 662–668.
- Pirasteh, G., Saidur, R., Rahman, S.M.A., Rahim, N.A., 2014a. A review on development of solar drying applications. Renew. Sustain. Energy Rev. 31, 133–148.
- Poblete, R., Painemal, O., 2020. Improvement of the solar drying process of sludge using thermal storage. J. Environ. Manage. 255.
- Pou, K.R.J., Tripathy, P.P., 2020. Process optimization of vacuum-assisted solar drying of crush, tear and curl (CTC) black tea. J. Biosyst. Eng. 45, 24–32.
- Prakash, O., Kumar, A., Laguri, V., 2018. Performance of Modified Greenhouse Dryer with Thermal Energy. energy repo 1–8. https://doi.org/10.1016/j. egyr.2016.06.003.
- Pranesh, V., Velraj, R., Christopher, S., Kumaresan, V., 2019. A 50 year review of basic and applied research in compound parabolic concentrating solar thermal collector for domestic and industrial applications. Sol. Energy 187, 293–340.
- Prasad, J., Prasad, A., Vijay, V.K., 2007. Studies on the Drying Characteristics of Zingiber Officinale Under Open Sun and Solar Biomass (Hybrid) Drying. Int. J. Green Energy 3, 79–89. https://doi.org/10.1080/01971520500439526.
- Pratoto, A., Daguenet, M., Zeghmati, B., 1998. A simplified technique for sizing solarassisted fixed-bed batch dryers : Application to granulated natural rubber. Energy Convers. Manag. 39, 963–971.
- Rabelo, S.C., Amezquita Fonseca, N.A., Andrade, R.R., Maciel Filho, R., Costa, A.C., 2011. Ethanol production from enzymatic hydrolysis of sugarcane bagasse pretreated with lime and alkaline hydrogen peroxide. Biomass and Bioenergy 35, 2600–2607. https://doi.org/10.1016/j.biombioe.2011.02.042.
- Rabha, D.K., Muthukumar, P., Somayaji, C., 2017. Energy and exergy analyses of the solar drying processes of ghost chilli pepper and ginger. Renew. Energy 105, 764–773. https://doi.org/10.1016/j.renene.2017.01.007.
- Rahman, M.S., 2006. Drying of Fish and Seafood, in: Handbook of Industrial Drying, Fourth Edition, Taylor & Francis Group, LLC. pp. 547–561. https://doi.org/10.1201/ 9781420017618.ch22.
- Rathore, N.S., Panwar, N.L., 2010. Experimental studies on hemi cylindrical walk-in type solar tunnel dryer for grape drying. Appl. Energy 87, 2764–2767. https://doi.org/ 10.1016/j.apenergy.2010.03.014.
- REN21, 2019. Renewables 2019. IEA (2019), IEA, Paris https//www.iea.org/reports/ renewables-2019.

- Salihoglu, N.K., Pinarli, V., Salihoglu, G., 2007. Solar drying in sludge management in Turkey. Renew. Energy 32, 1661–1675. https://doi.org/10.1016/j. renene.2006.08.001.
- Sekyere, C.K.K., Forson, F.K., Adam, F.W., 2016. Experimental investigation of the drying characteristics of a mixed mode natural convection solar crop dryer with back up heater. Renew. Energy 92, 532–542. https://doi.org/10.1016/j. renene.2016.02.020.
- Sharma, A., Dutta, A.K., Bora, M.K., Dutta, P.P., 2019. Study of energy management in a tea processing industry in Assam. India. AIP Conf, Proc, p. 2091.
- Sharma, A.K., Sharma, C., Mullick, S.C., Kandpal, T.C., 2016. Carbon mitigation potential of solar industrial process heating : paper industry in India. J. Clean. Prod. 112, 1683–1691. https://doi.org/10.1016/j.jclepro.2015.04.093.
- Shringi, V., Kothari, S., Panwar, N.L., 2014. Experimental investigation of drying of garlic clove in solar dryer using phase change material as energy storage. J. Therm. Anal. Calorim. 118, 29–41. https://doi.org/10.1007/s10973-014-3991-0.
- Singh, P., Kumar, A., Tekasakul, P., 2015. Applications of software in solar drying systems : A review. Renew. Sustain. Energy Rev. 51, 1326–1337.
- Singh, P., Shrivastava, V., Kumar, A., 2018. Recent developments in greenhouse solar drying : A review. Renew. Sustain. Energy Rev. 82 (3), 3250–3262. https://doi.org/ 10.1016/j.rser.2017.10.020.
- Singhal, U.M., Dixit, R., Arya, R.K., 2014. Drying of multilayer polymeric coatings, part I: An experimental study. Dry. Technol. 32, 1727–1740. https://doi.org/10.1080/ 07373937.2014.924526.

Sontakke, M.S., Salve, P.S.P., 2015. Solar Drying Technologies : A review 4, 29-35.

Subahana, K.R., Natarajan, R., 2016. Theoretical modelling and experimental investigation of the convective drying kinetics of biomass in an improved solar tunnel dryer. Biofuels 7269, 1–8. https://doi.org/10.1080/ 17597269.2015.1132372.

- Tagnamas, Z., Lamsyehe, H., Moussaoui, H., Bahammou, Y., Kouhila, M., Idlimam, A., Lamharrar, A., 2021. Energy and exergy analyses of carob pulp drying system based on a solar collector. Renew. Energy 163, 495–503. https://doi.org/10.1016/j. renene.2020.09.011.
- Tambunan, A.H., Manalu, L.P., Abdullah, K., 2010. Exergy Analysis on Simultaneous Charging and Discharging of Solar Thermal Storage for Drying Application. Dry. Technol. An Int. J. 28, 1107–1112. https://doi.org/10.1080/ 07373937.2010.506171.
- Tanwanichkul, B., Thepa, S., Rordprapat, W., 2013. Thermal modeling of the forced convection Sandwich Greenhouse drying system for rubber sheets. Energy Convers. Manae, 74, 511–523. https://doi.org/10.1016/i.encomman.2013.06.020.
- Thanaraj, T., Dharmasena, N.D.A., Samarajeewa, U., 2007. Original article Comparison of drying behaviour, quality and yield of copra processed in either a solar hybrid dryer on in an improved copra kiln. Int. J. Food Sci. Technol. 42, 125–132. https:// doi.org/10.1111/j.1365-2621.2006.01087.x.
- The International Energy Agency (IEA), 2018. U . S . Primary Energy Consumption by Source and Sector.
- Udomkun, P., Romuli, S., Schock, S., Mahayothee, B., Sartas, M., Wossen, T., Njukwe, E., Vanlauwe, B., Müller, J., 2020. Review of solar dryers for agricultural products in Asia and Africa: An innovation landscape approach. J. Environ. Manage. 268, 110730 https://doi.org/10.1016/j.jenvman.2020.110730.

Venkata Sai, P., Reddy, K.S., 2020. 4-E (Energy-Exergy-Environment-Economic) analyses of integrated solar powered jaggery production plant with different pan configurations. Sol. Energy 197, 126–143. https://doi.org/10.1016/j. solener.2019.12.026.

- Vijayaraj, B., Saravanan, R., 2008. Numerical Modeling of Moisture and Temperature Distribution within a Rectangular Bagasse Layer Undergoing Drying. Dry. Technol. 26, 749–758. https://doi.org/10.1080/07373930802046419.
- Xu, C., Li, X., Xu, G., Xin, T., Yang, Y., Liu, W., Wang, M., 2018. Energy, exergy and economic analyses of a novel solar-lignite hybrid power generation process using lignite pre-drying. Energy Convers. Manag. 170, 19–33. https://doi.org/10.1016/j. enconman.2018.05.078.
- Xu, C., Xin, T., Xu, G., Li, X., Liu, W., Yang, Y., 2017. Thermodynamic analysis of a novel solar-hybrid system for low-rank coal upgrading and power generation. Energy 141, 1737–1749. https://doi.org/10.1016/j.energy.2017.11.046.
- Yahya, M., Fahmi, H., Fudholi, A., Sopian, K., 2018. Performance and economic analyses on solar-assisted heat pump fluidised bed dryer integrated with biomass furnace for rice drying. Sol. Energy 174, 1058–1067. https://doi.org/10.1016/j. solener.2018.10.002.
- Yahya, M., Fudholi, A., Sopian, K., 2017. Energy and exergy analyses of solar-assisted fluidized bed drying integrated with biomass furnace. Renew. Energy 105, 22–29. https://doi.org/10.1016/j.renene.2016.12.049.
- Zachariah, R., Maatallah, T., Modi, A., 2020. Environmental and economic analysis of a photovoltaic assisted mixed mode solar dryer with thermal energy storage and exhaust air recirculation. Int. J. Energy Res. 1–13 https://doi.org/10.1002/er.5868.
- Zhu, T.T., Diao, Y.H., Zhao, Y.H., Wang, T.Y., Liu, J., 2018. A Comparative Investigation of Two Types of MHPA Flat-Plate Solar Air Collector Based on Exergy Analysis. J. Sol. Energy Eng. 139, 1–11. https://doi.org/10.1115/1.4037385.